Vol. 19, No. 1, 2024



swiss academies reports

swiss-academies.ch

RNA technologies: mechanisms of action, applications and forms of delivery

ABOUT THIS PUBLICATION

PUBLISHER AND CONTACT

Swiss Academy of Sciences (SCNAT) Forum for Genetic Research House of Academies • Laupenstrasse 7 • P.O. Box • 3001 Bern • Switzerland +41 31 306 93 36 • geneticresearch@scnat.ch • geneticresearch.ch • X @scnatCH

RECOMMENDED FORM OF CITATION

Kümin M, Eyer K, Hall J, Koch A, Pascolo S, Romeis J (2024) RNA technologies: mechanisms of action, applications and forms of delivery. Swiss Academies Reports 19 (1)

AUTHORS

Michael Kümin (Forum for Genetic Research, SCNAT) • Klaus Eyer (ETH Zurich) • Jonathan Hall (ETH Zurich) • Aline Koch (University of Regensburg) • Steve Pascolo (University of Zurich) • Jörg Romeis (Agroscope)

PROJECT MANAGEMENT

Franziska Oeschger (Forum for Genetic Research, SCNAT) • Michael Kümin (Forum for Genetic Research, SCNAT)

MONITORING GROUP FOEN AND SECB

Christoph Lüthi (FOEN) • Basil Gerber (FOEN) • Franziska Bosshard (FOEN) • Julia Link (SECB) • Elisabetta Peduzzi (SECB)

REVIEW

Khalid Amari Baba (Julius Kühn Institute) • Etienne Bucher (Agroscope) • Patrice de Werra (Swiss Society for Phytiatry) • Ursula Jenal (Jenal & Partners Biosafety Consulting)) • Rory Johnson (University College Dublin) • Roland Kölliker (Swiss Society of Agronomy) • Oliver Yves Martin (ETH Zurich) • Hans-Peter Meyer (Swiss Academy of Engineering Sciences SATW) • Oliver Mühlemann (University of Bern) • Barbara Rothen-Rutishauser (Adolphe Merkle Institute, University of Fribourg) • Adrian Rüegsegger (Foundation for Technology Assessment TA-SWISS), Florian Steiner (University of Geneva) • Christina Weinberg (Leipzig University)

PROOFREADING TRANSLATION

CVB INTERNATIONAL

LAYOUT Monique Borer

COVER PICTURE

 $\label{eq:def_Dr_Microbe} {\tt Dr_Microbe} - {\tt stock.adobe.com} \ ({\tt generated} \ {\tt with} \ {\tt AI})$

This report was written with the financial support of the Federal Office for the Environment and the Swiss Expert Committee for Biosafety. The SCNAT is solely responsible for the content.

1st edition, 2024

ISSN (online) 2297-184X

DOI: doi.org/10.5281/zenodo.10926908



RNA technologies: mechanisms of action, applications and forms of delivery

On behalf of the Federal Office for the Environment (FOEN) and the Swiss Expert Committee for Biosafety

Table of contents

1	Introduction					
1.1	Background					
1.2	Aim of the report					
1.3	Featured technologies					
1.4	Struct	Structure of the report				
2	Post-	transcrip	tional gene silencing	8		
2.1	Introduction					
	2.1.1	Antisen	ise oligonucleotides	8		
	2.1.2	RNA int	erference	8		
2.2	Therapeutic applications					
	2.2.1	Description of the procedures				
		2.2.1.1	Antisense oligonucleotides	9		
			Small interfering RNA (siRNA)			
			miRNA-based therapeutics			
		2.2.1.4	Chemical modifications			
			2.2.1.4.1 ASOs			
			2.2.1.4.2 siRNAs			
	2.2.2		f development			
		2.2.2.1	ASOs	. 11		
		2.2.2.2	siRNAs			
			2.2.2.2.1 Challenges			
	2.2.3		able effects			
		2.2.3.1	ASOs			
			2.2.3.1.1 Impurities			
			siRNAs			
2.3			n plant protection			
			Description of the procedure			
	2.3.2		tions			
			Insect pests			
			Viruses			
			Fungi and oomycetes			
			Bacteria			
	2.3.3		f development			
			Challenges			
	2.3.4		nded effects			
		2.3.4.1	In the target organism			
			2.3.4.1.1 Off-target effects			
			2.3.4.1.2 Resistances			
		2.3.4.2	Effects on non-target organisms			
			2.3.4.2.1 Stability in the environment			
			2.3.4.2.2 Sequence-specific effects			
			2.3.4.2.3 Sequence-unspecific effects			
		2.3.4.3	Contamination	.18		
-			la des			
3	mRNA technologies					
3.1	Introduction					
3.2	Description of the procedure					
3.3	арріі	cations		20		

	3.3.1 mRNA vaccines						
	3.3.2 Antibodies						
~ (3.3.3 mRNA-based protein replacement therapies						
3.4	State of development						
	3.4.1 Vaccines						
	3.4.1.1 Infectious diseases						
	3.4.1.2 Cancer						
	3.4.2 Antibodies						
	3.4.3 mRNA-based protein replacement therapies						
	3.4.4 Veterinary medicine						
	3.4.5 Challenges						
3.5	Undesirable effects	. 23					
4	RNA aptamers24						
4.1	Introduction						
4.2	Description of the procedure						
	4.2. Binding to target molecules	.24					
	4.2.2 Aptamers vs antibodies	.24					
	4.2.3 SELEX						
	4.2.4 Chemical modifications	. 25					
	4.2.4.1 Nuclease activity	. 25					
	4.2.4.2 Aptamer diversity	. 25					
	4.2.4.3 Renal filtration	. 25					
4.3	Applications	. 25					
	4.3.1 Therapies	. 25					
	4.3.1.1 Target inhibition	. 25					
	4.3.1.2 Targeted drug delivery	. 26					
	4.3.2 Diagnostics	. 27					
	4.3.3 Applications in agriculture	. 27					
4.4	Unintended changes and effects	. 27					
4.5	State of development	. 27					
5	Long non-coding RNAs	. 29					
5.1	Introduction	. 29					
5.2	Description of the procedure and applications	. 30					
5.3	State of development	. 30					
	5.3.1 Challenges	. 30					
5.4	Possible adverse effects	.30					
6	RNA-directed DNA methylation	31					
6.1	Introduction	31					
6.2	Description of the procedure						
6.3	Possible applications and state of development	. 32					
7	RNA activation						
7.1	Introduction	. 33					
7.2	Description of the procedure	. 33					
7.3	Applications and state of development						
	7.3.1 Challenges	.34					

8	Circular RNAs	35
8.1	Introduction	35
8.2	Description of the procedure and applications	36
8.3	State of development	36
9	Ribozymes	
9.1	Introduction	37
9.2	Description of the procedure	37
9.3	Possible applications and state of development	37
	9.3.1 Challenges	37
10	Delivery	38
10.1	Medical delivery methods	38
	10.1.1 Naked RNA	38
	10.1.2 Non-viral vectors	38
	10.1.2.1 Lipid nanoparticles and liposomes	38
	10.1.2.2 Polymer-based nanoparticles	39
	10.1.2.3 Cell-penetrating peptides	39
	10.1.3 Applications and state of development	39
	10.1.3.1 Challenges	40
	10.1.4 Undesirable effects	40
10.2	Delivery in plant protection	40
	10.2.1 Naked RNA	40
	10.2.2 Mini cells	41
	10.2.3 Nanoparticles	41
	10.2.3.1 Layered double hydroxide	41
	10.2.3.2 Carbon quantum dots (CQDs)	41
	10.2.3.3 Gold nanoparticles	41
	10.2.3.4 DNA nanostructures	41
	10.2.3.5 Chitosan nanoparticles (CNPs)	41
	10.2.3.6 Single-walled carbon nanotubes (SWCNT)	41
	10.2.4 State of development	42
Refe	rences	43
Glos	sary	48
	-	

List of figures

Figure 1:	Classification by mechanism of action	6
Figure 2:	RNA silencing	8
Figure 3:	Functionality of ASOs	10
Figure 4:	RNA modifications	12
Figure 5:	Mechanism of action of RNAi using an RNA spray as a crop protection agent	15
Figure 6:	How mRNA technologies work	22
	SELEX	
Figure 8:	Functional mechanisms of IncRNA	29
Figure 9:	RNA-mediated DNA methylation in Arabidopsis thaliana	31
Figure 10:	RNA activation	33
Figure 11:	circRNA-based medical applications	35
Figure 12:	Medical delivery methods	42

1 Introduction

1.1 Background

Ribonucleic acids (RNA) play an essential role in all cells of all organisms. Nevertheless, until recently, these molecules were overshadowed by the much better-known deoxyribonucleic acid (DNA). It was only the COVID-19 pandemic that brought RNAs in the form of mRNA vaccines into the spotlight: less than a year after the outbreak of the pandemic, mRNA vaccines against SARS-CoV-2 were available in most western countries. This was the result of decades of research into both the structure and function of RNAs and their use in therapy. In Switzerland, around 70% of the population was vaccinated against SARS-CoV-2 in April 2023, the vast majority with an mRNA vaccine.

RNAs occur naturally in all cells and fulfil a wide variety of functions. They play a prominent role in the production of proteins, for example by transferring genetic information from the DNA in the cell nucleus to the cytoplasm via messenger RNAs (mRNAs). However, RNAs also fulfil many regulatory functions in cells and can bind to a variety of target molecules. This makes them potentially suitable for a wide range of applications, particularly in medicine and agriculture. RNA technologies, especially for therapeutic applications, have been developed for over 30 years. The first RNA-based therapies were authorised at the turn of the century (see Section 2.2.2.1 ASOs). However, the technologies were not yet widely accepted at that time. Only in recent years have decisive technical advances been made, for example by stabilising RNA and introducing it efficiently into target cells. As a result, several new RNA therapies have been developed within a few years and public and industrial research into RNA technologies has intensified considerably (Mollocana-Lara et al., 2021). With the success of the mRNA vaccines against SARS-CoV-2, RNA technologies have now moved even more into focus. Various research groups and networks, e.g. the Swiss National Center of Competence in Research RNA & Disease (NCCR RNA & Disease), as well as private companies, are also working on the research and development of RNA technologies, particularly in the field of therapeutics.

1.2 Aim of the report

The aim of this report is to provide a technical and scientific overview of the various currently relevant RNA technologies. It provides information on the mechanisms of action of the various RNA-based approaches, their fields of application and products and their development status. The report is intended to help assess and discuss the benefits, risks and limitations of the various RNA technologies and their products. It is also intended to support the legal categorisation of RNA products.

1.3 Featured technologies

The report discusses various technologies based on RNA molecules. RNA stands for ribonucleic acid and consists of the sugar ribose and the four bases adenine, guanine, cytosine and uracil. Chemically, RNA differs from DNA (deoxyribonucleic acid) in that it contains the base uracil (U) instead of thymine (T) and that DNA lacks a hydroxyl group on the sugar-phosphate backbone compared to RNA. Unlike DNA, which is practically always doublestranded, RNA is often single-stranded. In contrast to DNA, RNA is also very susceptible to degradation by enzymes called ribonucleases, which are ubiquitous in organisms. In many RNA technologies, the RNA molecules are therefore chemically modified to increase their stability, among other things. These modified nucleic acids cannot be clearly classified as RNA or DNA if they do not have the hydroxyl group that distinguishes the two substances. There are also active substances that can consist of both RNA and DNA molecules, for example antisense oligonucleotides (ASOs) (see Section 2.1.1). RNA technologies are therefore not always clearly distinguishable from DNA technologies. Where appropriate, DNA-based methods are therefore occasionally mentioned in the report if their mechanism of action is the same as that of RNA technologies.

The report does not cover technologies that use RNA molecules to modify the genetic material of target cells. This includes in particular the CRISPR/Cas method for genome editing, which uses two RNA molecules in addition to the Cas enzyme. RNA viruses and virus-like vectors, such as replicons, are also not covered, as these differ significantly from the technologies presented here in various aspects, particularly with regard to biological safety.

The report focuses on RNA technologies that are being researched and developed with a view to applications in therapy and agriculture. It considers both technologies that are already established and have initial products on the market (such as mRNA vaccines) and approaches that are at an early stage of research and development. The RNA-based methods relevant to the report were identified on the basis of the specialist literature and assessments by experts and categorised into eight groups:

- 1. post-transcriptional gene silencing
- 2. mRNA technologies
- 3. RNA aptamers
- 4. long non-coding RNAs (lncRNA)
- 5. RNA-directed DNA methylation (RdDM)
- 6. RNA activation (RNAa)
- 7. circular RNAs (circRNAs)
- 8. ribozymes

The various RNA-based methods can differ in the type of RNA molecules used, the target molecules to which they bind and/or the mechanism of action (see Figure 1).

The distinction between the different methods is not always clear, and various categories and designations are also used in the literature. The grouping used here is therefore only one of several ways of categorising the different RNA technologies.

1.4 Structure of the report

The first chapters (Chapters 2–5) describe in detail four RNA technologies that are already established with products and/or are considered to have great potential for future applications in therapy or agriculture. The mechanism of action, possible applications and development status for each technology are described.

The relevance of the various methods was determined on the basis of the literature and the assessment of the experts involved (see imprint). Overall, post-transcriptional gene silencing (Chapter 2) and mRNAs (Chapter 3) are currently regarded as the two most relevant approaches. Both have already established themselves with the first therapeutic products and have great potential for further medical applications. Based on the post-transcriptional gene silencing approach, applications for agriculture are also being researched and developed. The first crop protection product has already been registered for authorisation in the USA. Aptamers (Chapter 4) are also an established technology with authorised products. However, no further authorisations have been granted in recent years and research has long been limited to niche applications. Long, non-coding RNAs (Chapter 5) are still at a relatively early stage of development. However, due to their diverse mechanisms of action, their potential is considered to be great, even if many challenges remain.

Chapters 6–9 provide a brief overview of four other RNA technologies that are also being researched and developed but are still at an early stage of development: RNA-dependent DNA methylation can regulate the activity of genes in plants by attaching chemical groups to the DNA; RNA activation causes increased gene activity by binding short RNAs to RNA or DNA molecules; circular RNAs are ring-shaped RNAs with various functions; ribozymes can catalyse chemical reactions.

Delivery methods are an important area of research and development for all RNA technologies in order to bring the RNA molecules specifically to the desired tissues and into the corresponding cells. Since RNA molecules are susceptible to degradation by ribonucleases and have difficulty penetrating cells, they need to be packaged for most applications, e.g. in nanoparticles or liposomes. The successful improvement of the formulation of these particles has been a crucial step that has helped newer RNA



Figure 1: Classification by mechanism of action.

The RNAs presented here act either by complementary binding to RNA or DNA molecules, binding to molecules via their three-dimensional structure or translating the encoded information. therapeutics, such as the mRNA vaccine against SARS-CoV-2, to achieve a breakthrough. Research into developing even more efficient and tissue-specific or cell-specific delivery methods is a very active and rapidly evolving field. The final chapter provides an overview of the currently most relevant delivery methods for therapeutic and agricultural applications.

Preparation of the report

The report was compiled by a group of six authors with expertise in the field of RNA technologies under the leadership of the SCNAT Forum for Genetic Research. The entire report was then reviewed by other specialists with different areas of expertise. Details of the people involved can be found in the imprint.

Categories of RNAs

RNAs can be divided into different groups in various ways and labelled accordingly.

They can be categorised into small RNAs (sRNA) and long RNAs (long RNAs) based on their **length**. Small RNAs have an approximate length of less than 200 nucleotides, long RNAs more than 200 nucleotides. Long RNAs mainly include mRNAs and long non-coding RNAs (lncRNAs).

Furthermore, the **structure** of RNAs can be labelled as either single-stranded (ssRNA) or double-stranded (dsRNA). Certain RNAs are also classified according to their **shape**, for example as circular RNA (circRNA) or hairpin-shaped RNAs (e.g. short hairpin RNA, shRNA).

Finally, RNAs can also be classified according to their **function**. At the highest level, a distinction is made between coding RNA and non-coding RNA (ncRNA). Coding RNAs contain the information for the production of a protein. ncRNAs, on the other hand, take on a variety of other cell functions, for example in the regulation of gene expression, the defence against viruses or the protection of the genome. ncRNAs are further differentiated according to their specific functions. These include, for example, small interfering RNAs (siRNAs), which bind and inhibit specific target RNAs, or ribozymes, which catalyse chemical reactions. However, there is a subgroup of RNAs that are translated and are nevertheless categorised as long non-coding RNAs.

2 Post-transcriptional gene silencing

2.1 Introduction

2.1.1 Antisense oligonucleotides

In 1978, researchers synthesised a small piece of DNA (also known as an oligonucleotide) that was complementary to a short RNA sequence of the Rous sarcoma virus. When the researchers added the piece of DNA to a cell culture, they observed that the virus could no longer spread in the cells. The synthesised DNA sequence had bound the complementary RNA sequence of the virus and thus prevented the virus from replicating. It was later shown that oligonucleotides can also bind the cell's own RNAs and thereby downregulate genes. This opened up the possibility of influencing viral or cellular RNAs for research purposes or for medical applications. Since then, this technology has been continuously developed to specifically regulate target genes in certain cells (Crooke et al., 2021).

2.1.2 RNA interference

In the 1990s, it was discovered that introducing doublestranded RNAs into a cell can also inhibit the expression of certain genes. It turned out that this mechanism, called RNA interference, occurs in almost all eukaryotic life forms and plays an important role in regulating gene expression, defending against viruses and protecting the genome (Svoboda, 2020).

Various types of short RNAs that can regulate genes occur in nature (Svoboda, 2020). They are all non-coding RNAs (ncRNA), which means that their genetic information is not used for the production of a protein. The two most common short, non-coding RNAs are small interfering RNAs (siRNA) and microRNAs (miRNA). Other short ncRNAs often have very specific functions, such as piwi-interacting RNAs (piRNA), which regulate socalled retrotransposons in the gametes. The various short non-coding RNAs differ in their length, origin and the type of target RNAs.

The 21–24 nucleotide (nt) long siRNAs are produced in nature by the cleavage of a long dsRNA into short pieces by the Dicer enzyme (Dicer-like (DCL) in plants). In vertebrates, the original double-stranded RNA (dsRNA) usually originates from a virus or virus-like element (e.g. a transposon). In plants and invertebrates, siRNAs can also be encoded in the genome and regulate their own genes (e.g. trans-acting siRNAs). In addition, siRNAs can be multiplied in plant cells by special polymerases and spread throughout the plant (Svoboda, 2020).



Figure 2: RNA silencing.

A) Basic process: non-coding RNA is cleaved into shorter RNA by Dicer and bound by an Argonaute protein (AGO). Together with AGO, the short RNA binds a sequence on the target RNA. The effect can be enhanced by an RNA-dependent RNA polymerase, which amplifies the dsRNA. B) RNA interference. Synthetic dsRNAs or dsRNAs from viruses or from the cell's own transcripts are converted into siRNAs by Dicer and dsRNA-binding proteins, which lead to enzymatic cleavage of the mRNA when they bind to the target mRNA. C) miRNA in mammals: in the cell nucleus, the pri-miRNA encoded in the genome is truncated to pre-miRNA by the Drosha and DGR8 proteins and then transported into the cytoplasm. There it is shortened to miRNA by Dicer and binds to the target mRNA together with AGO and other proteins. If sequence complementarity is not complete, translation is blocked and the mRNA is degraded (Svoboda, 2020). Natural miRNAs, on the other hand, are always encoded in the genome and are present after transcription as primary miRNAs (pri-miRNAs), which can be hundreds of nucleotides long. These pri-miRNAs are then processed into 70–80 nt long miRNA precursors (pre-miRNAs) and transported from the cell nucleus into the cytoplasm of the cell, where they are cut by Dicer into 17–24 nt long double-stranded miRNAs (Svoboda, 2020).

Both siRNAs and miRNAs are then bound by a protein from the Argonaute family. Subsequently, one of the RNA strands is removed and the resulting RNA-induced silencing complex (RISC), consisting of Argonaute and other proteins, and single-stranded siRNA or miRNA bind to a specific sequence of the target RNA. If the complementarity between the target RNA and the siRNA is high, the target mRNA is enzymatically cleaved by the RISC complex. This is the normal case with siRNAs, but can also occur with miRNAs, especially in plants. In animals, however, the complementarity of miRNAs with the target mRNAs is usually only partial, which means that the RISC complex binds the mRNA but does not cleave it. As a result, translation is blocked and the mRNA is often degraded (Svoboda, 2020).

Due to their higher complementarity, siRNAs are more specific because they can usually only bind one target RNA. miRNAs, on the other hand, can bind several mRNAs and thus inhibit the production of different proteins simultaneously (Svoboda, 2020).

While in plants both siRNAs and miRNAs play an important role in the defence against viruses and gene regulation, siRNAs are only of secondary importance in animals, especially mammals. This is also shown by the fact that the two types of dsRNA in plants are processed by a complex network of different protein variants. In mammals, on the other hand, they are processed in separate processes, whereby that of siRNAs is greatly reduced. Synthetic siRNAs that have been introduced into a mammalian cell are processed by the same factors as miRNAs (Svoboda, 2020).

Short double-stranded RNAs can inhibit not only translation but also transcription (see Chapter 6 'RNA-directed DNA methylation').

2.2 Therapeutic applications

The ability of ncRNA molecules to bind specifically to certain sequences of other RNAs and thereby influence protein synthesis can be utilised for therapeutic purposes in several ways. On the one hand, single-stranded or double-stranded RNA molecules can be produced that penetrate the affected cells and reduce the amount of a target RNA there. This can reduce the quantity of disease-causing proteins. On the other hand, the production of disease-preventing proteins can be increased by binding harmful miRNAs, which reduce the production of the corresponding protein. In addition to mRNAs, long non-coding RNAs can also be considered as targets for regulation (see Chapter 5 'Long non-coding RNAs').

miRNA-based therapeutics are not considered a separate class of molecules but are an umbrella term for siRNAs or ASOs that either bind to miRNAs or replace miRNAs (see 2.2.1.3 'miRNA-based therapeutics').

2.2.1 Description of the procedures

2.2.1.1 Antisense oligonucleotides

Antisense oligonucleotides (ASOs) are synthetic, singlestranded sequences with a length of 12–24 nucleotides. They can be DNA, RNA or artificial nucleotides – such as LNA (locked nucleic acids). Ribonucleotides are often modified and combined with artificial nucleotides in order to obtain oligonucleotides with the desired pharmacological properties (see Section 2.2.1.4 'Chemical modifications') (Crooke et al., 2021; Hall et al., 2022).

ASOs can act via different mechanisms. By binding to the target mRNA, some ASOs can initiate its cleavage by the ribonuclease H enzyme (RNAse H). This utilises a natural mechanism of the cell, which has the task of eliminating naturally occurring DNA-RNA duplexes, for example during transcription or viral infections. This occurs primarily with DNA-based, modified ASOs or so-called gapmers. Gapmers consist of a DNA sequence surrounded by modified oligonucleotides. ASOs can also be used to modify splicing. During splicing, certain areas are cut out of a precursor form of the mRNA, the pre-mRNA, and are therefore not translated into a protein during translation (see also Chapter 3 'mRNA technologies'). This process can be altered by ASOs in such a way that, for example, a stop codon is incorporated into the mRNA, which leads to an early termination of translation and degradation of the target RNA. Or a sequence with a harmful mutation can be skipped so that the mutation is not included in the final mRNA (see Figure 3).

In addition, modified single-stranded RNAs can act like siRNAs and – when incorporated into the RISC complex – lead to enzymatic cleavage of the target RNA (see Section 2.2.1.2 'Small interfering RNA (siRNA)') (Crooke et al., 2021).

However, ASOs can also upregulate gene expression, for example by binding to miRNAs and thus impairing their function (see Section 2.2.1.3 'miRNA-based therapeutics') or by increasing the translation rate by binding to certain regulatory sequences in the mRNA (Crooke et al., 2021).

The therapeutically less relevant mechanisms of ASOs include preventing translation of the target mRNA by blocking the ribosome or the cleavage of the so-called cap structure at the end of the mRNA. In addition, ASOs can change the length of the so-called poly(A) tail at the other end of the mRNA and thereby influence the stability of the mRNA (Crooke et al., 2021).

2.2.1.2 Small interfering RNA (siRNA)

Small interfering RNAs (siRNAs) are 19–24 nucleotidelong double-stranded RNA molecules. Together with several proteins, they form the RNA-induced silencing complex (RISC). In the RISC complex, one of the strands, called the leading strand, binds a complementary sequence on the target mRNA by base pairing. In most cases, this leads to enzymatic cleavage of the target mRNA (Svoboda, 2020).

Introducing siRNAs into a cell can be used to specifically degrade both coding and non-coding RNAs. This makes it possible to downregulate genes that cause disease due to a mutation or dysregulation. siRNAs, like ASOs, are characterised by the fact that they can bind any RNA whose sequence is known with a high affinity and specificity. Small biomolecules or antibodies that are intended to influence protein function, on the other hand, must be designed to fit correctly into a complex three-dimensional shape, which is difficult or impossible to achieve with many proteins. However, this higher binding capacity also makes it easier for siRNAs to bind to non-target molecules, which can lead to unintended (off-target) effects. In addition, siRNAs are relatively short-lived in the body. In order to improve their pharmacological properties, siRNAs are usually chemically modified for therapeutic applications (Hu et al., 2020) (see Section 2.2.1.4 'Chemical modifications').



Figure 3: Functionality of ASOs.

 ASOs containing deoxynucleotides (gapmers) bind a complementary sequence of an mRNA and recruit RNase H1, which blocks translation and degradation of the mRNA. 2) ASOs can influence where splicing factors (SFs) can bind by binding to complementary sequences in pre-mRNAs. This can lead to the inclusion or skipping of an exon. 3) They can lead to premature termination of translation and degradation of the mRNA. 4) They can block translation. 5) They can prevent the binding of miRNAs. (Y. Zhu et al., 2022) To date, four RNA interference (RNAi)-based therapeutics have been approved by the US Food and Drug Administration (FDA); others are in the clinical phase (see Section 2.2.2 'State of development').

2.2.1.3 miRNA-based therapeutics

There are two different strategies for miRNA-based therapeutics: miRNA antagonism and miRNA replacement therapies. The aim of miRNA antagonism (anti-miR) is to suppress or reduce the expression or function of a defective target miRNA. Antagomirs are short single-stranded non-coding RNAs (with a similar structure to ASOs), which bind to the target miRNAs and interfere with their function through complementary hybridisation and steric hindrance. In other words, they prevent the miRNA from binding to the natural target mRNA. In turn, miRNA replacement therapy aims to restore the expression or function of a miRNA. For example, this may be a miRNA that downregulates certain genes that promote the development of cancer. If this miRNA is downregulated, the risk of a tumour developing increases. In replacement therapy, a short, double-stranded RNA molecule (miRNA mimic) is synthesised that resembles the endogenous miRNA. This is then introduced into the cell, where it takes over the function of the natural miRNA (Traber & Yu, 2022). Unlike siRNAs, which are very specific, miRNAs often regulate the expression of several genes. This property makes it possible to use miRNA-based therapeutics to simultaneously influence several critical elements in deregulated cellular processes (Traber & Yu, 2022).

While some anti-miR therapies are already in the clinical phase, miRNA replacement therapies are only in the preclinical phase (see Section 2.3.3 'State of development').

2.2.1.4 Chemical modifications

As RNA is very unstable *in vivo* and can be rapidly degraded by enzymes (nucleases), it must be chemically modified for many medical applications in order to protect the phosphodiester backbone from nucleases. However, modifications can also have a negative effect on the ability to bind to the target molecule. The search for modifications that improve stability, binding capacity, cellular uptake and other pharmacological properties without reducing binding strength and specificity too much is one of the main focuses of medical research in this area.

2.2.1.4.1 ASOs

The most common modification in ASOs, the replacement of the natural phosphodiester bonds between the nucleotides with synthetic phosphothioate (PS) bonds, prevents cleavage by nucleases and facilitates cell entry for the oligonucleotide. In addition, the stronger binding to serum proteins helps the drug to spread in tissues and organs. The 2'-OH group of ribose, the only site where DNA and RNA differ chemically in the nucleotide chain, is often replaced by a 2'-O-methoxyethyl group (2'-MOE), a methyl group (2'-OMe) or an enclosed ethyl group (2'-cEt). Alternatively, the ribose moiety can be linked to LNAs (locked nucleic acids) with an additional C-O bridge to increase the binding affinity to the target RNA. The ribose sugar can also be replaced by a synthetic methylene morpholine ring (phosphorodiamidate Morpholino oligomers, PMOs, also known as Morpholinos), which prevents degradation by ribonucleases. To facilitate uptake in liver cells, the ASOs can be combined with N-acetylgalactosamine (GalNAc). In addition, there are a variety of different chemical modifications for adapting the oligonucleotides to the desired mode of action and route of delivery (Adachi et al., 2021; Hall et al., 2022). It is also possible to insert the sequence into a circular RNA, which is more stable than linear RNA and less likely to be recognised by the immune system (see Chapter 8 'Circular RNAs').

2.2.1.4.2 siRNAs

The first therapeutic siRNAs (e.g. Bevasiranib) were largely unmodified and were administered locally. However, they showed low efficacy and off-target effects. In addition, the innate immune response can be triggered. Therefore, many of the modifications that have proven successful in ASOs are also used for siRNAs, in particular, 2'-O-methyl (2'-OMe) and phosphothioate (PS) linkages, which prevent cleavage by nucleases, and N-acetylgalactosamine (GalNAc), which promotes uptake into certain liver cells (hepatocytes) (Hu et al., 2020).

2.2.2 State of development

2.2.2.1 ASOs

To date, eight ASO-based drugs have been approved by the US Food and Drug Administration (FDA) and four by the European Medicines Agency (EMA). The first, Fomivirsen (FDA: 1998, EMA: 1999, withdrawn in the EU in 2002 and in the US in 2006 for economic reasons) was used for the treatment of cytomegalovirus infections and was an unmodified DNA ASO that binds viral RNA. Eteplirsen (FDA: 2016, EMA: rejected), Golodirsen (FDA: 2019), Viltolarsen (FDA: 2020) and Casimersen (FDA: 2021) are Morpholino oligomers and Nusinersen (FDA: 2016, EMA: 2017) is a 2'-MOE-modified ASO. All of these lead to the skipping of a coding RNA segment (exons) which, if it contains a certain mutation, can cause Duchenne muscular dystrophy. Mipomersen (FDA: 2013, EMA: rejected) for the treatment of homozygous familial hypercholesterolaemia, Inotersen (FDA: 2018, EMA: 2018) for the treatment of hereditary transthyretin-mediated amyloidosis and Volanesorsen (EMA: 2019) for the treatment of familial chylomicronaemia syndrome are 2'-MOE- modified ASOs that, when binding to mRNA molecules, cause their degradation by RNAse H (Crooke et al., 2021). With the exception of Fomivirsen, which was used for viral eye inflammation, all drugs approved to date are aimed at the treatment of rare hereditary diseases; they are therefore used in an area in which there is little competition from small molecule drugs and biopharmaceuticals (e.g. monoclonal antibodies) and therefore a high-price policy is possible (Hall et al., 2022).

In 2020, seven ASO-based drugs were tested in clinical phase III and 31 in phase II. These include drugs for diseases of the nervous system; muscular, cardiovascular, inflammatory and metabolic diseases; eye diseases; various cancers and infectious diseases (Crooke et al., 2021).

Another potential application of ASOs is in the treatment of very rare diseases through personalised medicine. Milasen was developed specifically for the treatment of a single patient by identifying her specific genetic defect and synthesising a PS-MOE oligonucleotide to correct the splicing defect (Hall et al., 2022).

2.2.2.2 siRNAs

Four RNAi-based therapeutics have already been approved in Europe and the US: Patisiran (FDA & EMA, 2018; Swissmedic, 2019), Givosiran (FDA, 2019; EMA, 2020; Swissmedic, 2021), Lumasiran (FDA & EMA, 2020;

Swissmedic 2021) and Inclisiran (EMA, 2020; FDA & Swissmedic, 2021).

Patisiran (Onpattro®) was the first RNAi-based drug to be approved by the FDA in 2018 for the treatment of familial amyloid polyneuropathy. In this disease, amyloid plaques form from misfolded transthyretin (TTR), a protein that is mainly produced in liver cells. Patisiran binds to the TTR mRNA and thereby reduces the formation of amyloid plaques. Givosiran (Givlaari®) is authorised for the treatment of acute hepatic porphyria (AHP). In AHP, toxic substances accumulate in the cell due to overproduction of the protein aminolevulinate synthase 1 (ALAS1). Givosiran binds to the ALAS1 mRNA and thereby reduces the amount of protein. Inclisiran (Leqvio®) is approved for the treatment of familial hypercholesterolaemia and clinical atherosclerotic cardiovascular disease and reduces the enzyme proprotein convertase subtilisin/kexin type 9 (PCSK9) in the liver (Hall et al., 2022; Traber & Yu, 2022). Lumasiran (Oxlumo®) is used to treat primary hyperoxaluria type 1. In this disease, oxalate is deposited in various organs. Lumasiran reduces the enzyme glycolate oxidase (GO), which produces a precursor of oxalate in the liver, thereby lowering the amount of oxalate in the body (Traber & Yu, 2022).

Patisiran consists of two multi-methylated RNA strands enclosed in a lipid nanoparticle that facilitates uptake



Figure 4: RNA modifications. A) Modifications to the 2'-position of the ribose B) Modifications to the base (Feng et al., 2021).

into liver cells (see Section 10.1.2.1 'Lipid nanoparticles and liposomes'). The second-generation siRNA drugs – Givosiran, Lumisiran and Inclisiran – no longer contain natural nucleotides and are bound to N-acetylgalactosamine (GalNAc) for increased uptake into certain liver cells (so-called hepatocytes) (Hall et al., 2022).

2.2.2.2.1 Challenges

Therapeutic siRNAs must enter the cytoplasm and/or the cell nucleus of the target cells in order to become active. To do this, they have to overcome a number of obstacles. They must remain stable in the blood, i.e. they must not be degraded enzymatically, and they must evade recognition by the immune system. They must then accumulate in the desired tissue and be transported through the cell membrane (Y. Zhu et al., 2022).

In oligonucleotide drug development, attempts are being made to improve transport into cell types and tissues other than the liver by means of chemical modifications (see Section 2.2.1.4 'Chemical modifications'). On the other hand, various delivery vehicles - for example made of lipids, lipid-like materials, polymers, peptides, exosomes and nanoparticles - are being developed (see Section 10.1 'Medical delivery methods') (Hu et al., 2020). The conjugation of GalNAc ligands to siRNA is a tried-and-tested method for introducing siRNAs into liver cells. Methods for targeting other cell types have not yet been established but are being intensively researched. Therefore, siRNAs have so far only been delivered intravenously into liver cells (Crooke et al., 2021). Delivery systems that can deliver siRNAs specifically into the central nervous system, eyes, kidney cancer cells and epithelial cells are now at an advanced stage of development (Hu et al., 2020).

2.2.3 Undesirable effects

2.2.3.1 ASOs

The undesirable effects that can occur with ASOs depend heavily on their chemical composition, as this can influence the mechanism of action and the interactions with other molecules (see Section 2.2.3.1 'Chemical modifications'). A distinction must also be made between off-target effects, which are caused by complementary base pairing outside the target sequence, and non-specific effects, such as immune reactions or interactions with cellular proteins (Jason et al., 2004).

The frequently used phosphothionate (PS) modification in ASOs can induce a non-specific pro-inflammatory response by binding to immune receptors. PS-ASOs can activate both the complement system and the innate immune system and have the potential to have an immunotoxic and cytotoxic effect. However, this risk can be greatly reduced by certain modifications, such as 2'-MOE. On the one hand, the modifications can prevent recognition by immune receptors and, on the other hand, reduce the amount of molecules required for therapy through better pharmacological properties (Crooke et al., 2021). For some PS-ASOs containing 2'-MOE, cEt or LNA modifications, cell-toxic effects have been observed, probably caused by the interaction with cellular proteins. However, the addition of 2'-O-methyl (2'-OMe) greatly attenuated this effect (Shen et al., 2019). In contrast, actual off-target effects due to non-specific binding of RNase H1 are rarely expected if the target sequence has been carefully selected, as high complementarity is required for cleavage (Crooke et al., 2021).

2.2.3.1.1 Impurities

In addition to organic and inorganic impurities and residues of solvents, which can also occur during the production of other therapeutic agents such as low-molecular compounds, product-specific impurities also arise in oligonucleotide therapeutics.

In addition to the desired main oligonucleotides, a complex mixture of oligonucleotide by-products is formed during their synthesis. These by-products differ from the main oligonucleotide either in the nucleotide sequence, structure or individual bonds between the nucleotides. Oligonucleotide by-products that are structurally identical to the main products or only contain naturally occurring structural elements can, in principle, trigger off-target effects and, under certain circumstances, inflammatory reactions. However, they are usually only present in quantities that are not pharmacologically relevant. By-products containing structural elements that are neither present in the main oligonucleotide nor in naturally occurring nucleotides are difficult to assess in terms of their toxicological properties. Capaldi and colleagues recommend chemically characterising oligonucleotide impurities and testing them in animals if a certain qualification threshold is exceeded, which depends on the structural elements present in the impurity. They believe that the results of these tests can be transferred to chemically similar oligonucleotides so that the tests do not have to be repeated for each new oligonucleotide from the same class (Capaldi et al., 2017).

2.2.3.2 siRNAs

siRNAs that are bound in the RISC complex should only bind to a specific target RNA. However, the complex tolerates certain errors in complementarity, which is why sequences that are similar to the target sequence can also be bound. The probability of such an off-target effect can be reduced by bioinformatic analyses and an appropriate design of the siRNAs. Another source of off-target effects is the second strand of the siRNA, the sense strand, which should not bind to the target mRNA. It can happen that instead of the intended antisense strand, the sense strand remains in the RISC complex and binds an unwanted RNA. However, this can be prevented by chemically modifying the sense strand so that it is more difficult to load into the RISC complex (Hu et al., 2020).

As with ASOs, siRNAs also harbour the risk of activating the innate immune system. Attempts are also being made here to reduce the probability of an immune reaction through chemical modifications (Hu et al., 2020).

2.3 Applications in plant protection

2.3.1 Description of the procedure

Crops are threatened by a variety of viruses, fungi, bacteria and other microorganisms. Animal pests can also cause major crop losses. In agriculture, these pathogens and pests can be kept in check by suitable soil cultivation, diverse crop rotations and resistant varieties, but they are also combated with biological and synthetic plant protection products. However, these have a relatively unspecific effect, i.e. they also affect non-target organisms and can have a negative impact on ecosystems, biodiversity and other protected goods. Due to the sequence-specific mechanism of RNAi, dsRNAs can be used to downregulate essential genes of pathogens and harmful organisms relatively specifically, thereby weakening or killing them (OECD, 2020; K. Y. Zhu & Palli, 2020; Zotti et al., 2018).

The double-stranded RNAs (dsRNAs) that trigger the silencing effect can either be expressed endogenously (i.e. in the plant) by incorporating corresponding DNA sequences into the plant genome (by producing genetically modified plants), or they can be synthesised in bacteria or cell-free systems and applied externally (exogenously) (Zotti et al., 2018). The endogenous expression and formation of dsRNA is also known as host-induced gene silencing (HIGS). It has the advantage that dsRNAs are present in sufficient quantities in the plant and can either act against pathogens or be taken up by harmful organisms (Koch & Petschenka, 2022). However, plants that express dsRNAs endogenously are transgenic organisms, as foreign DNA is incorporated into the genome. They are therefore not the subject of this report and are not discussed in detail below, but only mentioned where they contribute to the understanding of the individual applications and their development status.

Exogenous dsRNA, on the other hand, is either applied to the plant surface by spraying – in this case it is referred to as spray-induced gene silencing (SIGS) – or introduced directly into the plant interior, e.g. by injection or mechanical inoculation (see Figure 5). For many applications, systemic distribution of the dsRNA in the plant is required (see Section 2.3.3.1 'Challenges'). Systemic distribution is facilitated by injecting or mechanically inoculating the dsRNA into the plant, as the dsRNA does not have to penetrate the plant surface first. However, these methods are currently only suitable for use in the laboratory, partly due to the high manual effort involved (Rank and Koch 2021).

2.3.2 Applications

2.3.2.1 Insect pests

Experiments have shown that RNAi in insects can be triggered by dsRNAs when ingested. By expressing the corresponding dsRNAs from the crop plant itself or spraying them onto it, it is possible for the dsRNA to be taken up by insect pests during feeding and to down-regulate certain genes that are essential for the development and physiological processes of these insects. This approach can be used to produce sequence-specific insecticides adapted to specific target insects, which act very selectively and can be quickly customised (see as an example Bachman et al., 2013, 2016). Extensive RNAi screens have now been carried out on a number of pests to identify essential genes that could be suitable targets for this method (Koch & Petschenka, 2022; Zotti et al., 2018).

However, the effectiveness of such dsRNAs has proven to be highly variable. While certain insect orders such as Orthoptera (grasshoppers), Blattodea (cockroaches and termites) and most Coleoptera (beetles) show a strong RNAi response following the uptake of dsRNAs, this response is significantly weaker in Lepidoptera (butterflies and moths) and Diptera (flies). In the Hemiptera (plant lice, cicadas and bugs), the response varies greatly (Huvenne & Smagghe, 2010; Koch & Petschenka, 2022; OECD, 2020).

The effectiveness of the dsRNA is influenced by various factors. These include the stability of the RNA, the efficiency of uptake of the dsRNA by the insects and the expression level of the genes required for the RNAi process (see Section 2.1.2 'RNA interference'). The speed at which dsRNAs are processed or degraded by nucleases also has an influence on their effectiveness. RNAs can spread systemically in some insect species, while they remain localised in others (S. Liu et al., 2020). In some insect species that reproduce asexually (especially aphids), RNAi activity with decreasing strength has also been observed in subsequent generations. This is probably due to the direct transfer of dsRNA from the mother to her offspring while she is carrying it (Abdellatef et al., 2015).

When using dsRNA in spray form, it is important to distinguish between two main exposure routes due to the different feeding behaviour of insect pests. RNA active ingredients that are particularly effective against leaf-feeding insects, such as the Colorado potato beetle, must be adap-ted so that they exhibit high surface stability and adhesion. dsRNAs against sucking insects, on the other hand, must be absorbed into the leaf interior (apoplast/ intercellular space) and into the vascular tissue (phloem/xylem). One difficulty is protecting the RNA on the plant from environmental influences such as UV radiation or rain. Both in the plant and in the insect, the RNA must remain stable until it can bind the target mRNA (Christiaens et al., 2020; Koch & Petschenka, 2022). There is also evidence that the dsRNA can be absorbed to a small extent via the outer skin (cuticle) of insects (Christiaens et al., 2018; Hoang et al., 2022; Huang et al., 2019; Romeis & Widmer, 2020).

RNA sprays have already been successfully used under laboratory conditions against various harmful insects and arachnids. These include the ladybird *(Henosepilachna vigintioctopunctata)*, the Colorado potato beetle *(Leptinotarsa decemlineata)* and the common pollen beetle *(Brassicogethes aeneus)*. However, there is still hardly any reliable data from field trials (Koch & Petschenka, 2022) (see Section 2.3.3 'State of development').

2.3.2.2 Viruses

RNAi is an important natural defence mechanism of plants against viruses (see Section 2.1.2 'RNA interference'). It is therefore not surprising that dsRNAs have proven to be an effective means of controlling viral pathogens. Both transgene-based and exogenous RNAi methods have shown a very high efficacy of 90% against viruses (Koch & Wassenegger, 2021). It has also been shown that exogenous dsRNAs can spread systemically in the plant, which is crucial for their effectiveness against viruses. However, many of these applications require complex pre-treatment of the plants, for example by mechanically damaging the leaves, whereby a dsRNA abrasive solution is manually rubbed onto the leaves. For this reason, the search is currently on for possible formulations that facilitate cellular uptake (see Section 2.3.3.1 'Challenges') (Rank & Koch, 2021).

2.3.2.3 Fungi and oomycetes

Various HIGS-based studies have shown that the RNAi approach is also effective against phytopathogenic fungi and oomycetes (a group of organisms that are physiologically similar to fungi but not closely related to them). dsRNA sprays have been successfully used in the laboratory against Fusarium graminearum, the pathogen that causes fusarium ear blight (Höfle et al., 2020; Koch et al., 2016, 2019; Werner et al., 2020). However, the ability to take up dsRNAs varies considerably between the fungal and oomycete species tested. Necrotrophic fungi, which feed on dead plant cells, appear to react much more sensitively to exogenous RNAs than biotrophic fungi, which can only take up RNAs from the cell interior of their host plants. It is therefore assumed that, depending on their lifestyle, fungi can take up dsRNAs from the extracellular as well as from the intercellular and intracellular space of the plant. As with other RNA-based applications, the



Figure 5: Mechanism of action of RNAi using an RNA spray as a crop protection agent.

After entering the plant, a long dsRNA is cleaved by DCL into shorter siRNAs, which are taken up by the target organism and lead to the degradation of a target mRNA (Koch & Petschenka, 2022).

often insufficient stability and limited cellular uptake of RNA is an obstacle to the development of anti-fungal RNA sprays (see Section 2.3.3.1 'Challenges').

2.3.2.4 Bacteria

Most bacterial diseases in plants are difficult to combat. The use of antibiotics for direct control is undesirable due to the risk of resistance development and the associated problems in human and veterinary medicine. Therefore, often only preventive measures against bacterial plant diseases are possible. There are currently only a few studies on the use of HIGS against bacterial pathogens, in particular against plant tumours caused by *Agrobacterium tumefaciens* (Escobar et al., 2001) and against *Pectobacterium carotovorum* (Mahmoudi & Soleimani, 2019). No trials with SIGS-based methods are known.

2.3.3 State of development

For a long time, RNAi-based crop protection focused on genetically modified, transgenic plants that express dsRNAs endogenously. Their effectiveness has been demonstrated in many studies, and several plant varieties are already ready for the market (Rank & Koch, 2021). The maize MON87411 (SmartStax® PRO) is already authorised for trade (e.g. EU) and cultivation (e.g. USA, China). In addition to several Cry proteins from the bacterium *Bacillus thuringiensis*, it also expresses a dsRNA that makes it resistant to the corn rootworm (S. Liu et al., 2020).

RNA sprays were long considered uneconomical because the cost of producing synthetic dsRNAs was very high. Approximately 20 g/ha are required for application in the field (Koch & Petschenka, 2022). This is low compared with conventional pesticides, but prices of several thousand dollars per gram make an RNA-based product significantly more expensive. However, thanks to new cell-free synthesis processes, the price has now fallen to less than one dollar per gram. This has increased interest in the development of RNAi-based pesticides (Rank and Koch 2021; Hoang et al. 2022).

To date, extensive field trials have been carried out in the European Union with the dsRNA-based insecticide Ledprona from GreenLight Biosciences[™] (product name Calantha[™]) against the Colorado potato beetle *(Leptinotarsa decemlineata)*. The effect on the mortality of the beetle larvae was comparable to that of the conventional insecticides spinosad and chlorantraniliprole (Pallis et al., 2022). In 2021, GreenLight Biosciences[™] submitted an application to the US Environmental Protection Agency (EPA) to register Ledprona as a new active ingredient (EPA, 2021) and received a tolerance exemption in spring. In December 2023, provisional authorisation was granted for three years, which will subsequently be reviewed (EPA 2023). An application for registration in the EU is also expected.

After acquiring the spray-based RNAi platform Bio-Direct[®] to control the bee mite *Varroa destructor* from Bayer/Monsanto, GreenLight BiosciencesTM further developed it into a ready-to-use product (pollinator). This was successfully tested in several large-scale trials, which led to the submission of a product label for registration with the EPA (de Schutter et al., 2022).

In addition, GreenLight BiosciencesTM has already developed further dsRNA-based plant protection concepts against the fungi *Erysiphe necator, Botrytis, Sclerotinia* and *Fusarium* and successfully evaluated them in the field (although not yet published).

2.3.3.1 Challenges

To be effective, dsRNAs must remain stable for a long time under very diverse conditions. dsRNA that is applied to plant surfaces is exposed to rain and UV radiation. dsRNA that is to act systemically throughout the plant must be taken up by the plant and transported to distant tissues. Depending on the application, it must also reach the inside of the plant cells. After uptake by an insect, it must pass through the digestive tract, be absorbed into the cells and exert its effect there. However, RNA is a relatively unstable, short-lived molecule. This means that the sprayed dsRNA often does not reach the target cells or does not reach them in sufficient quantities. As with the medical applications of RNAi, protection against nucleases and extreme pH values as well as the targeted transport of the RNA pose major challenges (Hoang et al., 2022; Rank & Koch, 2021).

Solutions based on nanomaterials are being sought for many of these problems. dsRNA attached to clay nanosheets (BioClay™/LDH) is less easy to wash off (see Section 10.2.3.1 'Layered double hydroxide'). 'Single-walled carbon nanotubes' (SWCNT) and carbon quantum dot nanoparticles could help to introduce dsRNA into plant cells and protect it from enzyme activity (see Section 10.2.3.6 'Single-walled carbon nanotubes (SWCNT)' and Section 10.2.3.2 'Carbon quantum dots'). To improve stability, the RNA can be chemically modified, as in medical applications. The fact that these nanomaterials and chemical modifications can influence the stability and chemical interactions in the plant, the target organisms and in the environment also leads to new potential risks that should be investigated prior to authorisation (Christiaens et al. 2020; Rank and Koch 2021; Hoang et al. 2022; S. Liu et al. 2020: Romeis and Widmer 2020). Another difficulty is that it is not yet fully understood

how dsRNAs are taken up into the leaves and cells and which factors are involved in RNAi activation (Hoang et al., 2022; Rank & Koch, 2021).

2.3.4 Unintended effects

As RNA spray applications have so far mainly been tested against insect pests, only the known unintended effects of these applications are described in the following sections.

2.3.4.1 In the target organism

2.3.4.1.1 Off-target effects

For effective silencing, relatively long dsRNAs are usually used, which are cleaved by the Dicer into several short siRNAs. As the siRNAs do not have to be fully complementary to the target sequence during base pairing, it is likely that not only the target mRNA is bound, but also other mRNAs that contain a similar sequence (OECD, 2020). In the target organism, however, such off-target effects are unproblematic, as they lead at most to further damage to this organism (S. Liu et al., 2020; OECD, 2020).

2.3.4.1.2 Resistances

There are concerns that single base mutations in the target gene present in the insect population could reduce the effectiveness of RNAi and lead to the evolution of resistance (Scott et al., 2013). However, this risk is low if a pool of different siRNAs or long dsRNAs are used to compensate for the non-binding of individual siRNAs. It is also speculated that mutations in molecules that play a role in the RNAi process and viruses that code for RNAi suppressors could give certain insects an evolutionary advantage (Scott et al., 2013). However, as insects are dependent on a functioning RNAi mechanism, it is unlikely that disruption of this mechanism by mutations or viruses would lead to a selective advantage.

However, the greater danger seems to lie in the fact that insect populations become resistant by preventing the uptake of dsRNA via the gut. This mechanism has been demonstrated for resistant populations of the western corn rootworm *Diabrotica virgifera virgifera* (Khajuria et al., 2018) and the Colorado potato beetle (Mishra et al., 2021) which were selected under artificial laboratory conditions. These populations would therefore also be resistant to dsRNA targeting other genes.

2.3.4.2 Effects on non-target organisms

As with all plant protection products, the potential impact of RNAi-based plant protection products on non-target organisms – including humans as users and consumers – of an ecosystem must be clarified (P. M. Bachman et al., 2016; Christiaens et al., 2018; Romeis & Widmer, 2020). The decisive factor here is whether other organisms come into contact with biologically active dsRNAs and take them up. This depends, among other things, on how long these molecules remain stable in the environment. A distinction must also be made between effects that are specific to the RNA sequence and non-specific effects such as immune reactions (Romeis and Widmer 2020).

2.3.4.2.1 Stability in the environment

In general, naked (i.e. unformulated) RNA does not remain stable for long in the environment. Several studies have shown that naked dsRNAs could no longer be detected 48 hours after being applied to clayey soils and after seven days in aquatic systems with different sediment types (P. Bachman et al., 2020). Naked dsRNA that had been sprayed onto the leaves of potatoes was difficult to wash off after drying and proved to be significantly more UV-resistant than on glass surfaces. Even four weeks after spraying, this dsRNA was still biologically active. However, it is unclear whether the dsRNAs actually remained on the surface or whether they had been taken up into the plant (S. Liu et al., 2020).

If formulated dsRNA is used, the formulations (see Section 10.2 'Delivery in plant protection') significantly influence their stability. For example, dsRNAs bound to BioClay/LDH could still be detected on the tobacco plant *(Nicotiana tabacum)* 30 days after spraying, while naked dsRNAs were barely detectable after 20 days (Mitter et al., 2017). It was also shown that carbon quantum dots protect dsRNAs from degradation by RNase A for at least 30 days (Wang et al., 2023).

2.3.4.2.2 Sequence-specific effects

When non-target arthropods come into contact with biologically active dsRNA, it is possible that they take it up and process it using their RNAi machinery. The likelihood of this happening depends strongly on various factors, such as the species or the length of the RNA (Romeis & Widmer, 2020). If the non-target arthropod expresses an mRNA whose sequence is sufficiently similar to the target sequence, unintentional gene silencing may occur. The probability of this can be reduced by choosing suitable sequences for the dsRNAs. When a larger number of insect genomes are decoded and available in the future, potential sequence matches in non-target organisms should be easier to predict bioinformatically. However, purely bioinformatic predictions may overestimate the effects. It is also important to consider whether the non-target organisms actually take up the dsRNAs and whether they have a harmful effect on them. Furthermore, the effects of siRNAs can be positively or negatively influenced by the presence of other naturally occurring siRNAs (S. Liu et al., 2020).

2.3.4.2.3 Sequence-unspecific effects

It is known that dsRNAs can trigger an immune response in mammals, including humans. Whether the uptake of dsRNAs via food and feed plants previously sprayed with RNA has an effect on the immune system is still unclear. However, early reports of the presence of dietary dsRNA in mammalian body fluids and tissues have not vet been confirmed (Kang et al., 2017). This also seems rather unlikely due to the various physical and biochemical barriers that would have to be overcome. Nevertheless, the probability of such undesirable effects increases if the dsRNAs are modified (see Section 2.2.1.4 'Chemical modifications') in order to increase their stability and cellular uptake (S. Liu et al., 2020). Non-specific immune responses are also observed in non-target insects when high doses of dsRNA are ingested or injected (Christiaens et al., 2018; OECD, 2020; Romeis & Widmer, 2020; Weinstock et al., 2006).

2.3.4.3 Contamination

dsRNA for use in RNA sprays can be produced either by chemical synthesis, by in vitro transcription, in bacterial cell cultures or in cell-free systems. In chemical synthesis, in vitro transcription and most bacterial methods, a single strand is synthesised first and then a complementary strand is added. Using RNA bacteriophages – viruses that infect bacteria – dsRNAs can also be synthesised directly. Contamination can occur with all these methods if the dsRNA is not carefully purified. However, this can be prevented by quality control measures. In addition, dsRNA can now be produced in large quantities using cell-free systems at low cost and without residues (Dietz-Pfeilstetter, 2021).

3 mRNA technologies

3.1 Introduction

Messenger RNAs (mRNAs) are single-stranded ribonucleic acids that carry the information for the production of a protein. The protein is produced in the cytoplasm of the cell. The possibility of introducing artificially produced mRNAs into a cell and producing practically any desired protein opens up many potential applications.

The best established mRNA-based application to date are the mRNA vaccines. These introduce an mRNA molecule containing the genetic information for a specific protein of a pathogen into body cells, where the protein is subsequently produced. This protein is then recognised as foreign by the immune system and stimulates an immune response that ultimately leads to immunisation against the pathogen. Compared to conventional vaccines, mRNA vaccines can be produced very quickly using cell-free processes. They can also be quickly adapted to mutations in the pathogen. The method is also suitable for pathogens against which no conventional vaccines are currently available. The method can even be used against cancer cells (Chaudhary et al., 2021).

Another application of synthetic mRNAs is passive immunotherapy. Antibodies have proven to be extremely effective in the treatment of a wide range of diseases – from autoimmune diseases to infectious diseases and various types of cancer. However, their production is very complex and they have to be administered repeatedly. Administering the antibody sequence as mRNAs, which are subsequently translated into proteins in the body, promises simpler production and a longer-lasting effect compared with administration as a protein (Schlake et al., 2019).

An mRNA-based method for replacing or supplementing endogenous proteins that are not produced in the required quantity or have lost their function is also being developed. Here, the corresponding gene product is inserted into the cells as mRNA, where the replacement protein is produced, which then replaces or supplements the endogenous protein (Qin et al., 2022).

This report does not deal with the possibility of modifying the genome using synthetic mRNAs. This involves introducing an mRNA molecule into the cell, which then expresses a protein that can cleave DNA or change bases (genome editing). These include Cas enzymes, TALEN or zinc finger nucleases.

3.2 Description of the procedure

Natural mRNAs fulfil an essential function in the production of proteins in the cell. They are produced in the cell nucleus during transcription by converting the genetic information on the DNA into a precursor of the mRNA, called pre-mRNA. The pre-mRNA is then processed. The resulting mRNA is transported from the cell nucleus into the cytoplasm, where it is read by molecular complexes (ribosomes). Through this process, called translation, various amino acids are assembled into a specific protein according to the sequence information contained in the mRNA (Qin et al., 2022).

During processing in the cell nucleus, certain sequences known as introns are cut out of the pre-mRNA so that only the regions known as exons remain. This step is known as splicing. By cutting different sequences from the premRNA (alternative splicing), different mRNAs can be created from one pre-mRNA molecule. In addition, a structure called a cap is attached to one end (5'-end) of the pre-mRNA and a series of adenine nucleotides - called poly(A) tails - are attached to the other end (3'-end). These protect the RNA from enzymatic degradation by nucleases. They are also a prerequisite for transport out of the cell nucleus and for translation. The sequence that contains the blueprint for the protein - called the open reading frame (ORF) - is flanked by two shorter sequences that are not translated into a protein. These sequences are called untranslated regions (UTR) and are important for the stability and correct function of the mRNA. In particular, they contain binding sites for proteins that influence the stability and transport of the mRNA and initiate translation. In addition, miRNAs can bind in the 3'-UTR and thereby regulate the stability and translation of the mRNA (see Section 2.1.2 'RNA interference') (Qin et al., 2022).

For medical applications, the open reading frame together with the two non-translated regions is first synthesised as DNA, which is then transcribed. The poly(A) tail is often synthesised together with the open reading frame but can also be added to the mRNA by a specialised enzyme. The cap structure can be added either during transcription or later with enzymes. The synthetic mRNA is then introduced into the somatic cells, where it is translated into a protein in the cytoplasm of the cell (Qin et al., 2022).

There are various ways in which mRNA can be introduced into body cells. The simplest way is to inject 'naked' mRNA – i.e. an mRNA molecule without a coating – into the muscle, skin or a tumour, for example. However, mRNA, like all RNAs, is unstable and is quickly degraded by enzymes, often before it can develop its primary effect in the target cell. For this reason, various delivery systems have been and are being developed that can efficiently introduce mRNAs into the desired cells. Liposomes or lipid-based vectors such as lipid nanoparticles, which are composed of natural and artificial lipids, are currently the most widely used. In contrast to liposomes, lipid nanoparticles have a core of lipids and molecules with lipid-like properties in which the active ingredients are encapsulated. Research is also being carried out on polymer- or peptide-based nanoparticles or lipid-polymer hybrid nanoparticles. The section that codes for the protein can also be incorporated into the genome of a virus, which then introduces the genetic information into the cell (see Section 3.4.5 'Challenges' and Section 10.1 'Medical delivery methods') (Gupta et al., 2021).

3.3 Applications

3.3.1 mRNA vaccines

Conventional vaccines contain an attenuated or killed pathogen – or parts of it – which is recognised by the immune system after vaccination and thus leads to immunisation against the pathogen (Krammer, 2020). This procedure has proven to be very successful for a number of infectious diseases, such as smallpox or poliomyelitis. However, the method has also been shown to have limited or no effect on many pathogens. This is particularly the case when these are able to mutate quickly and thus evade the immune response. As the development of new conventional vaccines takes a relatively long time, a reaction to a new or mutated pathogen can only occur with a time delay.

In mRNA vaccines, no pathogens are administered, only the blueprint for certain proteins of the pathogen. These proteins, known as antigens, are then produced in the body's cells through translation of the mRNA and recognised as foreign by the immune system (see Figure 6).

mRNA vaccines have various advantages over conventional vaccines. They can be produced, scaled up and customised very quickly in a cell-free process, which makes it possible to react quickly to mutations. An mRNA can also carry the information of several proteins simultaneously, and it does not integrate into the genome, which is located in the cell nucleus. As the proteins are produced in the body's own cells, the glycosylation also corresponds to that of natural proteins. In addition, mRNA vaccines have the potential to activate the immune system not only against pathogens, but also against cancer cells (Chaudhary et al., 2021).

Antigen-presenting cells, such as dendritic cells, play an important role in recognising antigens and the subsequent immune response. They take up antigens and present them to the B and T cells of the immune system. Instead of injecting the mRNA into the muscle, it is therefore also possible to remove dendritic cells from the patient and insert the mRNA directly into these cells. The dendritic cells are then injected under the skin, where they trigger the immune response. This approach is much more complex than the direct injection of mRNAs, but has proven to be potentially effective, especially in vaccinations against cancer cells (Qin et al., 2022).

A rough distinction can be made between three categories of mRNA vaccines: 1) non-nucleoside-modified mRNAs, 2) nucleoside-modified mRNAs and 3) self-amplifying non-nucleoside-modified mRNAs. Non-nucleoside-modified mRNAs contain only the four canonical bases A, C, G and U. In nucleoside-modified mRNAs, one or more canonical bases have been replaced by modified bases to prevent the mRNA from being recognised by the innate immune system and to improve translation. These are mainly pseudouridine, 1-methylpseudouridine or 5-methylcytidine. Self-amplifying mRNAs differ from conventional mRNAs in that they not only contain the genetic information for the antigen, but also for several proteins that are needed by viruses to replicate their RNA. This allows the mRNA to replicate itself after being introduced into the cell. Therefore, very little initial mRNA is sufficient to achieve immunisation in this process. In addition, dsRNA structures are formed during amplification, which are recognised by the innate immune system and thus enhance the effect of the vaccine (Pardi et al., 2018).

3.3.2 Antibodies

For some years now, antibody-based therapies have been among the most promising medical interventions against a number of diseases that were previously difficult to treat. They have proven to be effective against viral and bacterial infections, autoimmune diseases and various types of cancer. This form of therapy uses so-called monoclonal antibodies that specifically bind to a certain section (epitope) of a target molecule, thereby blocking it in a targeted manner. The antibodies are usually produced in cell lines from mice or humans that have come into contact with the antigen. Alternatively, there are also socalled recombinant monoclonal antibodies. These can also be produced in vitro without prior immunisation and therefore more quickly. However, these production methods are very time-consuming and the purified antibodies must then be constantly stored and transported under refrigeration until they are administered (Lu et al., 2020). This time-consuming and expensive production process can be avoided by introducing only the genetic information in the form of mRNA into the body cells and producing a certain amount of antibodies there for a short time, which can then perform their function in the body (Qin et al., 2022).

3.3.3 mRNA-based protein replacement therapies

In mRNA replacement therapies, cellular proteins that are not produced in sufficient quantities or are not regulated correctly are replaced or supplemented. This is done by introducing the mRNA of the corresponding protein into the cell and expressing it. Replacement therapies are particularly suitable for the treatment of rare genetic metabolic diseases and hereditary diseases such as muscular dystrophies or cystic fibrosis (Xiao et al., 2022).

3.4 State of development

3.4.1 Vaccines

The possibility of using mRNAs as a vaccine was considered more than 30 years ago. The first promising trials were carried out in animals in the early 1990s and in humans in the early 2000s (Pascolo, 2021). However, despite these early successes and all the potential benefits, this technology was long considered to be unpromising. This was mainly due to the fact that the mRNA molecules were considered too unstable to be used effectively as medical devices. A further difficulty was achieving adequate stimulation of the immune system. During the SARS-CoV-2 pandemic, mRNA vaccines finally established themselves as the most promising approach within a short space of time and proved their effectiveness. Not least because of this success, research into mRNA vaccines against other pathogens and cancer cells has also increased significantly in recent years (Chaudhary et al., 2021).



Figure 6: How mRNA technologies work.

mRNAs are packaged in nanoparticles and taken up into the cell by endocytosis.

The mRNA is released from the endosome and the nanoparticle is dissolved. The mRNA is translated in the cytoplasm and the resulting protein either remains in the cytosol, is integrated into the cell membrane or is secreted (Bhat et al., 2021).

3.4.1.1 Infectious diseases

So far, mRNA vaccines have only been authorised against the SARS-CoV-2 coronavirus. The first two mRNA vaccines received emergency use authorisation from the US Food and Drug Administration (FDA) in 2020 and were also approved in the EU and Switzerland shortly afterwards. Both the Comirnaty® vaccine developed by Bio-NTech[™] and the Spikevax[®] vaccine from Moderna[®] contain the mRNA of the spike glycoprotein of SARS-CoV-2. The mRNAs were embedded in lipid nanoparticles and the uridine bases were replaced by N1-methylpseudouridine. Since then, several billion vaccine doses have been administered worldwide (Qin et al., 2022). Vaccines adapted to new virus variants from these manufacturers have already been authorised or are still in the clinical test phase. As of February 2024, in addition to the vaccines from BioNTech[™] and Moderna[®], the self-amplifying non-modified mRNA HGCO19 from Gennovo in India, ARCoV (PLA Academy of Military Science / Walvax Biotechnology® / Suzhou Abogen Biosciences™) in Indonesia and SYS6006 (CSPC Pharmaceutical Group™) in China have been approved. Four others are in clinical phase III and six in phase II. Vaccines against cytomegaloviruses, influenza viruses and the human respiratory syncytial virus (RSV) are also in clinical phase III. Vaccinations against a number of other viruses, including the human immunodeficiency virus (HIV), rabies and Eppstein-Barr, as well as against various bacteria and parasites are still in the preclinical or early clinical phase (vfa, 2023).

3.4.1.2 Cancer

mRNA vaccines for cancer treatment have been investigated in preclinical and clinical studies and have shown promising results in various types of cancer. To date, nine clinical trials with mRNA vaccines against cancer cells have been completed. 17 additional phase I and II trials with lipid-coated mRNA are ongoing or in the process of recruiting subjects. In July 2023, Merck and Moderna announced the start of a Phase III trial of their personalised mRNA-4157 vaccine against melanoma (Carvalho, 2023). mRNA-based vaccinations with dendritic cells are already further advanced in some cases: among the 19 planned or ongoing trials, two are in Phase III (Lorentzen et al., 2022).

3.4.2 Antibodies

Most mRNA-based antibody therapies are still in the preclinical phase. Only one antibody against the chikungunya virus is currently undergoing clinical testing. Antibodies against the Zika virus, HIV and human RSV have been successfully tested in mice. The monoclonal antibody trastuzumab, which has been approved against certain types of breast and stomach cancer, was successfully administered to mice as mRNA (Qin et al., 2022).

3.4.3 mRNA-based protein replacement therapies

Intensive research is currently being conducted into protein replacement therapies. In particular, therapies are being sought for heart and lung diseases, haematopoietic disorders, orthopaedic diseases, neurological diseases, metabolic diseases and cancer. However, the vast majority of these therapies still have preclinical status. The only two mRNAs that are in the clinical phase code for the vascular endothelial growth factor (VEGF) to regrow blood vessels in the heart and for the cell surface protein 'cystic fibrosis transmembrane conductance regulator' (CFTR) for the treatment of cystic fibrosis (Qin et al., 2022).

3.4.4 Veterinary medicine

Although many of the diseases against which mRNA vaccines are being developed can be transmitted by animals, there are as yet no applications of mRNA technologies in veterinary medicine. Unlike in human medicine, no mRNA vaccines have been approved for the vaccination of domestic and zoo animals against SARS-CoV-2, but protein vaccines and attenuated viruses (Sharun et al., 2021). An experimental vaccine against rabies is currently under development and has already been successfully tested in mice and dogs (Li et al., 2022). The progress made in the development of mRNA vaccines for humans and the successful trials with various vaccines in laboratory animals suggest that this technology will also be increasingly used in veterinary medicine in the future (Aida et al., 2021).

3.4.5 Challenges

Transporting the mRNA to and into the desired cells remains a major challenge. mRNA – like all single-stranded nucleic acids – is negatively charged and therefore has difficulty penetrating the negatively charged cell membrane. It is also very unstable in cells and is degraded within a few hours. Even if it has successfully penetrated the cell, there is a high probability that it will be trapped in an endosome, a membrane-enclosed organelle, and thus deactivated. Suitable delivery systems consisting of liposomes or polymers protect the mRNA from degradation, promote uptake by the cell and allow the mRNA to escape from the endosomes. However, these systems must not be harmful and must not trigger an immune response. In this way, mRNAs can now be reliably and specifically introduced into various liver cells (hepatocytes, Kupffer cells and endothelial cells). Lipid nanoparticles, which are composed of various lipids with different properties, are most frequently used for this purpose. Nanoparticles made of biodegradable polymers and cationic nanoemulsions can also be used (see Section 10.1 'Medical delivery methods') (Qin et al., 2022).

Non-endogenous mRNAs, in particular by-products of in vitro production, which may be present in the preparations (such as dsRNA), are recognised by various receptors of the innate immune system. On the one hand, this can hinder protein production and thus the effectiveness of immunisation. On the other hand, with mRNA vaccines, activation of the innate immune system is also an important factor for the activation and maturation of dendritic cells, which ensure effective and long-lasting immunity. The challenge is therefore to optimise the immunostimulatory properties of the mRNA in such a way that a reaction of the immune system is triggered without negatively affecting translation and without causing severe side effects. The choice of transport system can also influence the immune response. For example, there are lipid nanoparticles with cyclic lipids that trigger a specific immune response (Pardi et al., 2018). When administered intramuscularly, this activation of the immune system has been achieved without serious negative side effects in both nucleoside-modified and non-modified COVID-19 vaccines (Barbier et al., 2022).

The previously authorised mRNA vaccines must be stored deep-frozen at -90 °C to -60 °C and only have a relatively limited shelf life. Even if the shelf life has now been improved, the storage requirements restrict the areas of application. Here, too, it is expected that stability at higher room temperatures can be improved by means of improved delivery methods, such as lyophilised liposomes (Barbier et al., 2022).

3.5 Undesirable effects

Before the coronavirus pandemic, most mRNA technologies had only been tested on animals, and many of the clinical trials are still at an early stage. After the outbreak of the pandemic, a lot of resources were invested globally in vaccine research and development and the approval processes were accelerated, which enabled clinical trials and subsequent approval of the vaccines within a short period of time. An important aspect in the development of mRNA agents concerns the immune response: with mRNA vaccines, this must be optimised so that it is strong enough for immunisation while keeping side effects to a minimum. In contrast, mRNA-based antibodies and protein replacement therapies require the lowest pos-

sible immune response in order to minimise side effects (Barbier et al., 2022). Serious side effects were rarely observed in the context of vaccination against SARS-CoV-2. However, more cases of myocarditis, Bell's palsy, cerebral venous and sinus thrombosis, Guillain-Barré syndrome and a number of other disorders of the vascular and immune system have been recorded. As these also occur more frequently after infection with SARS-CoV-2, it is speculated that the spike protein encoded on the mRNA is responsible for this (Trougakos et al., 2022). Anaphylactic reactions have also occurred. It is assumed that these are due to the presence of the polymer polyethylene (PEG) in the lipid envelope (Chaudhary et al., 2021). New formulations (liposomes, LNP, polyplexes) of mRNA (modified or unmodified) and alternative injection sites (e.g. subcutaneous) are being tested to further improve mRNA vaccines.

4 RNA aptamers

4.1 Introduction

As early as 1967, it was shown that RNA coding for a protein, which in turn replicates the RNA, can evolve in a cell-free medium through random mutations and subsequent selection and thus be optimised for specific requirements (Mills et al., 1967). In 1990, a method called SELEX (systematic evolution of ligands by exponential enrichment) was developed (see Section 4.2.3 'SELEX'). Under this method, a series of selection steps can be used to generate RNAs that bind very specifically to a molecule thanks to their three-dimensional structure (Tuerk & Gold, 1990). In the early 2000s, natural RNAs (riboswitches) were discovered in bacteria that recognise certain metabolites and regulate metabolism in this way (Nudler & Mironov, 2004).

4.2 Description of the procedure

RNA aptamers are short (10–120 nt), non-coding, singlestranded RNA sequences (RNA oligonucleotides). Unlike non-coding RNAs in post-transcriptional silencing (see Section 2.1.2 'RNA interference'), they do not bind to other ribonucleic acids through base pairing but recognise and bind a specific target molecule through their threedimensional configuration, similar to monoclonal antibodies. This can either activate or inactivate the target molecule. In addition to RNA aptamers, there are also DNA aptamers and peptide aptamers. In the following sections, aptamers always refer to RNA aptamers.

4.2.1 Binding to target molecules

The aptamers bind to their target molecules through van der Waals forces, hydrogen bonds, electrostatic interactions, stacking of flat aromatic ligands and shape complementarity. Small target molecules are integrated into the aptamer, while large target molecules, such as surface proteins, have the aptamer inserted into the structure of the molecule (Kohlberger & Gadermaier, 2021).

4.2.2 Aptamers vs antibodies

Aptamers have a similar function to antibodies but have certain advantages over them that make them particularly interesting for therapeutic and diagnostic applications. RNA aptamers can fold back into their functional structure even after several denaturations (i.e. the loss of

their structure). As a result, they remain stable for several years even at room temperature under sterile conditions. Antibodies, on the other hand, are much more sensitive to temperature: they have to be stored cold (2-8 °C) and denature irreversibly at excessively high temperatures. Aptamers can be chemically synthesised. In addition, an in-vitro SELEX process usually only takes 2-8 weeks. This makes them cheap to produce and there are hardly any differences in quality between different batches. They can also be chemically modified relatively easily in this process (see Section 4.2.4 'Chemical modifications'). While the production of antibodies requires a strong immune response against the target molecule, aptamers can be optimised for a wide range of targets, from small molecules to cells, and hardly trigger an immune response when unmodified. This means that aptamers can also be obtained against molecules that are evolutionarily conserved and therefore do not trigger an immune response in animals. Aptamers are significantly smaller than antibodies and can penetrate many tissues better than antibodies. They have a high affinity and specificity towards the target molecule (Chen et al., 2015). In contrast, the affinity of antibodies depends on the number of binding sites (epitopes) on the antigen, and different antibodies can bind the same antigen (Kohlberger & Gadermaier, 2021).

4.2.3 SELEX

Aptamers that bind to specific molecules are selected using a process called SELEX. The target molecule is incubated together with 1014-1016 random singlestranded RNA oligonucleotides (called the aptamer library). These oligonucleotides usually consist of 40-100 nucleotides with random regions in the centre and constant sequences at the ends with binding sites for amplification by the polymerase chain reaction (PCR). The non-binding oligonucleotides are then washed away, while the binding oligonucleotides are reversetranscribed, i.e. converted into DNA, and then amplified by PCR. These DNA oligonucleotides are then transcribed and the resulting RNA oligonucleotides are used as a library for the next selection run. In several cycles, the aptamers that have a particularly high affinity and specificity for the target molecule are filtered out. At the end of the process, they are finally sequenced. In addition, aptamers that bind an undesired molecule can also be selected (counter selection). No prior knowledge of the target molecule is required for the SELEX process (Kohlberger & Gadermaier, 2021).

SELEX is often combined with other methods to optimise the specificity for certain target molecules or the efficiency of the process (e.g. Capillary Electrophoresis-SELEX, Microfluidic-SELEX or Atomic Force Microscopy-SELEX) (Y. Zhang et al., 2019).

One further development that has attracted particular attention is Cell-SELEX. Cell-SELEX is used to select aptamers that bind to surface proteins of living cells *in vitro*. The method is based on the fact that the target cells (e.g. tumour cells) have a specific expression pattern of the surface proteins. Aptamer selection is first carried out on a negative control (usually healthy cells). The unbound aptamers are then used as a pool for SELEX with the target cell line. Aptamers can also be specifically selected that are taken up by the target cells when they bind a surface protein (cell-internalisation SELEX) (Sola et al., 2020).

For in vivo SELEX, the aptamer libraries are injected into cancerous or pathogen-infected animals, usually mice, and then those aptamers that have bound to the affected tissue are amplified (Sola et al., 2020).

4.2.4 Chemical modifications

Aptamers can be modified to make them more specific and more affine for the target molecules and optimised for in vivo applications. They are therefore very rarely used unmodified in therapeutic applications (see Section 2.2.1.4 'Chemical modifications').

4.2.4.1 Nuclease activity

Unmodified aptamers often remain active in vivo for only a few minutes because they are very quickly digested by nucleases. In order to extend their half-life in the body, most aptamers are chemically modified so that they are no longer recognised by nucleases.

Aptamers can be chemically modified during selection (In-SELEX) or after selection (Post-SELEX). In-SELEX has the advantage of ensuring that the RNA remains functional despite modification. However, special polymerases are required for amplification because the modified RNA is not recognised by natural polymerases. Pegaptanib (Macugen®), a drug against macular degeneration and the first authorised aptamer therapeutic, was modified in this way. Post-SELEX, the nucleotides at the ends of the aptamer (5' or 3' position) are usually modified (Chandola & Neerathilingam, 2019).

Another way to avoid hydrolysis by nucleases is to use L-enantiomers of the ribonucleotides (called L-RNA aptamers or Spiegelmers). Here, a mirrored version of the target molecule is synthesised. This L-molecule is used as the target molecule in the SELEX process. The D-oligonucleotide that binds best is then sequenced and synthesised with L-nucleotides. The resulting L-aptamer should bind the natural D-molecule as strongly as the D-aptamer binds the L-molecule. Since natural nucleases can only bind D-oligonucleotides, L-aptamers are protected from hydrolysis (Vater & Klussmann, 2015).

4.2.4.2 Aptamer diversity

Another reason for chemical modifications is the increase in oligonucleotide diversity, which increases the probability that an aptamer has a high affinity to the target molecule. This can be achieved either via the addition of hydrophobic side chains (SOMAmers), via non-natural nucleotides or via glycosylation (Byun, 2021).

4.2.4.3 Renal filtration

Due to their small size, aptamers are quickly filtered out of the blood by the kidneys and excreted. This can be prevented by attaching large molecules – such as polyethylene glycol (PEG), liposomes, proteins, cholesterol or various nanomaterials (Kohlberger & Gadermaier, 2021).

4.3 Applications

4.3.1 Therapies

4.3.1.1 Target inhibition

Aptamers can bind to a target molecule and have either an activating (agonistic ligand) or inactivating (antagonistic ligand) effect. An aptamer that binds and deactivates the HIV Tat protein was developed as early as 1990. Since then, the catalogue of agonistic and antagonistic aptamers has grown steadily. In 2004, the first aptamer therapeutic (Pegaptanib) was approved for the treatment of age-related wet macular degeneration, a serious eye disease. This aptamer binds a specific isoform of vascular endothelial growth factor (VEGF) and thus prevents it from binding to the VEGF receptor. In all nine RNA aptamer drugs that have been clinically tested to date (Pegaptanib, EYE001, ARC19499, ARC1905, REG1, Nox-E36, Nox-A12, Nox-H94 and BT200), the aptamer binds a signalling molecule (Byun, 2021).

However, aptamers are also being developed that can prevent a virus from entering a cell. To do this, they bind to proteins that are necessary for attachment to the cell. They can also prevent the replication cycle of the virus by binding to proteins that are essential for replication (e.g. capsid proteins). Furthermore, they can compete with viral mRNA for an RNA polymerase binding site (Zou et al., 2019). An RNA aptamer-based drug against SARS-CoV-2 is also under development, which contains several RNA



Figure 7: SELEX.

First, a library of aptamers is synthesised. The aptamers are then brought into contact with the target molecule. The aptamers that bind the target most strongly are then selected. These are amplified and used for the next run until the binding is strong and specific enough. The resulting aptamers are sequenced (Kohlberger & Gadermaier, 2021).

aptamers that are attached to a gold nanoparticle and bind the cell surface protein ACE2 at various sites. This prevents the virus from binding to ACE2 and thus entering the cell (Sun et al., 2021, 2022).

4.3.1.2 Targeted drug delivery

One of the greatest challenges in the development of cell-specific or tissue-specific therapeutics is ensuring that the active substances reach the target cells and are taken up by them. Cell-SELEX can be used to select aptamers that bind surface proteins typical for certain cell types (see Section 4.2.3 'SELEX') and are taken up by the cell after binding. However, these methods are still at an early stage of development. There are two ways in which aptamers can fulfil this function:

Aptamers can be conjugated with a drug (aptamer-drug conjugates (ApDC)) or complexed. In ApDCs, the two mol-

ecules are connected by a linker molecule. This can be selected so that it becomes unstable under certain chemical conditions or can be cut by an enzyme, thus releasing the drug. This should make it possible to use the drug in a specific cell compartment or in the acidic environment of a tumour. ApDCs are being tested to deliver cytotoxins or radioisotopes specifically into cancer cells and thereby improve chemotherapy and radiotherapy. ApDCs are also being developed that modulate the immune system and could be used for immunotherapy (G. Zhu & Chen, 2018).

Aptamers can also be linked to liposomes that transport the active ingredient. When the aptamer is taken up into the cell, the liposome fuses with the cell membrane and releases the cargo molecule (see Section 10.1.2.1 'Lipid nanoparticles and liposomes'). Finally, aptamers can also guide viral vectors to a specific target (Chandola & Neerathilingam, 2019).

4.3.2 Diagnostics

Due to their similarity to antibodies, aptamers have the potential to replace them in various diagnostic applications (ELISA, affinity chromatography, etc.). Their high stability, flexibility and affinity make them particularly suitable for patient-oriented laboratory diagnostics (point-of-care instruments) and test kits for analyses in field research (Y. Zhang et al., 2019).

Using SELEX with whole pathogens or purified proteins as targets, aptamers have been selected that target a range of different bacteria (Saad & Faucher, 2021), viruses (e.g. SARS-CoV) (Kim & Lee, 2021) and parasites (e.g. *Tryptanosoma cruzi*) (Ospina-Villa et al., 2020). Cell-SELEX can also be used to develop new cancer biomarkers (Yan & Levy, 2018).

The most widely used aptamer-based biosensors (aptasensors) are so-called duplexed aptamers (DA), which are composed of two hybridised aptamers. One of the two aptamers can bind the target molecule as well as the other aptamer. When this happens, the aptamers separate and an optical or electrochemical signal is generated that can be measured (Munzar et al., 2019).

With a few exceptions, all of these technologies are in the development phase and are not yet ready for the market (see Section 4.5 'State of development').

4.3.3 Applications in agriculture

There are as yet no applications in plants. However, initial experiments with RNA aptamers are underway in basic research. For example, it has been shown that RNA aptamers can be expressed as transgenes in plant cells and can inhibit protein function (Abdeeva et al., 2019).

4.4 Unintended changes and effects

Aptamers have a short half-life in the body and a high specificity. No off-target effects due to binding to nontarget molecules have been observed to date. There are also no known interactions with other drugs. In toxicological studies, no activation of the immune or complement system has been observed with unmodified aptamers to date (Kovacevic et al., 2018). However, chemical modifications or binding to other large molecules (conjugations) can trigger an immunogenic or cytotoxic effect. For example, LNA (locked nucleic acids) – nucleic acids in which the sugar portion is chemically modified in a non-natural way – have shown strong hepatotoxicity. 2'-fluoropyrimidine-modified RNA aptamers lead to cytotoxicity and induce the expression of interferon-beta in human cancer cells in vitro. PEG, which is added to many aptamers to achieve a longer half-life in the blood, can trigger allergic reactions in rare cases (Chandola & Neerathilingam, 2019). In Phase IIb clinical trials of the aptamers pegnivacogin and anivamersen conjugated with PEG, one per cent of subjects experienced allergic reactions to PEG, and clinical trials with the anticoagulator Reg1 were discontinued due to severe allergic reactions (Povsic et al., 2016).

Because the expression levels of the surface proteins are unknown in Cell-SELEX, it is possible that aptamers are selected against certain surface proteins that are also frequently present on healthy cells and that these are therefore also targeted. This risk can be minimised by several rounds of negative selection (Chandola & Neerathilingam, 2019). If the aptamers are bound to liposomes, additional liposome-specific adverse effects may occur (see Section 10.1.2.1 'Lipid nanoparticles and liposomes').

4.5 State of development

Aptamers are used relatively widely in basic research. However, only a few aptamer-based drugs have made it into the clinical phase. Pegaptanib (Macugen®), the only RNA aptamer drug available on the market to date, received FDA approval for the treatment of macular degeneration in 2004. After a successful market launch, demand quickly declined due to competing drugs and it became a niche product. Since then, no aptamer-based drug has reached the market. Especially after the failure of the anticoagulant Reg1 in phase III trials, only a few aptamer drugs have reached the clinical phase. There are various reasons for this. The pharmacokinetic properties are relatively difficult to control. Aptamers can degrade, be excreted or be involved in metabolic processes, which can strongly influence the duration of the effect. To prevent degradation by nucleases and rapid excretion via the kidneys, aptamers are chemically modified, but this can have a negative impact on tolerability (see Section 2.3.4 'Unintended effects'). Another important reason is that aptamer binding to the target molecule and therapeutic effects are often not reproducible, which reduces confidence in the technology (Yan & Levy, 2018). The selection of aptamers is highly dependent on the conditions during the SELEX process. Even small differences in pH, the temperature of the buffer solution or the duration of the incubation can influence the selection. The conditions in the SELEX process also differ greatly from those in the natural environment of the molecule. In particular, the presence of ions or molecules with which the aptamer interacts electrostatically can strongly influence the structure, binding affinity and specificity. Therefore, a high binding affinity in vitro does not

guarantee that a similarly high affinity will be achieved *in vivo* (Byun, 2021).

In recent years, the use of aptamers for the targeted delivery of drugs has sparked interest (see Section 4.3.1.2 'Targeted drug delivery'). However, the aptamer-nanoparticle compounds are rapidly degraded *in vivo*. It is assumed that the nanoparticles are recognised by the immune system, that the nanoparticles aggregate or that the aptamers are cut enzymatically (Byun, 2021).

Aptamer-based instruments for diagnostics and environmental analyses (e.g. detection systems for microbes in water samples) are of great interest to both research and industry because aptasensors are relatively easy, quick and cheap to produce. However, one obstacle to their use in diagnostics is the lack of standardised protocols. Aptamers generated by the same process against the same target molecule can differ in their primary structures, binding affinities, specificities and other chemical parameters. Therefore, a protocol developed for one aptamer cannot be used for another. However, standardised kits and protocols - based on well-characterised aptamers - are essential for use in human diagnostics (Byun, 2021). For this reason, aptamer biosensors have so far mainly been used in basic research. No RNA aptamer biosensor has yet been approved for clinical use. In Indonesia, however, the DNA aptamer-based SARS-CoV-2 diagnostic kit 'Aptamex' has been on the market since 2022, and the company SomaLogic offers a SOMAmerbased proteome assay for non-clinical research with the 'SomaScan' platform (Grand View Research, 2023).

5 Long non-coding RNAs

5.1 Introduction

Long non-coding RNAs (lncRNAs) are all RNAs that are more than 200 nucleotides long and do not code for a protein. This broad definition means that this class includes a very diverse collection of transcripts. As of February 2024, more than 20000 lncRNAs are known in the human genome (Frankish et al. 2021). Although no biological function is yet known for most of these lncRNAs, it is becoming increasingly clear that some lncRNAs play an important role in the regulation of various cellular processes (Esposito et al., 2022; S. J. Liu et al., 2017; S. Zhu et al., 2016).

IncRNAs exhibit a broad spectrum of possible interaction mechanisms because, thanks to their length, they not only bind other ribonucleic acids by base pairing but can also form various three-dimensional structures (see Figure 8). In this way, they are able to form three-dimensional structures similar to siRNAs or miRNAs (see Chapter 2 'Post-transcriptional gene silencing'), in order to bind certain mRNAs by base pairing and influence their stability or translation into proteins. However, they can also bind complementary DNA sequences, forming either a so-called triplex structure with both DNA strands or a loop known as an R-loop. Finally, they can also bind to certain proteins due to their three-dimensional structure (Mercer et al., 2022; Statello et al., 2020).

LncRNAs are formed in a similar way to mRNAs. The reading of DNA by a protein known as RNA polymerase produces a transcript, which is usually spliced and receives a cap structure at one end and usually a poly(A) tail at the other end (see Section 3.2 'Description of the procedure'). Unlike mRNAs, however, not all lncRNAs are transported from the cell nucleus into the cytoplasm (Statello et al., 2020).



Figure 8: Functional mechanisms of IncRNA.

A) An IncRNA with multiple protein binding sites binds proteins. B) An IncRNA binds an enzyme as a co-factor and prevents the binding of a functional co-factor. C) An IncRNA binds an mRNA and influences its stability. D) An IncRNA binds an mRNA and influences its translation.
E) An IncRNA interacts with a DNA double strand. F) An IncRNA binds a DNA strand and thereby forms an R-loop (Mercer et al., 2022).

lncRNAs remaining in the cell nucleus can interact with the DNA or with proteins involved in transcription. In this way, they activate or suppress the transcription of certain genes. Many of the genes known to date that are regulated in this way are important for different developmental phases, such as certain phases of embryonic development in animals or flowering in plants. In some cases, they cause a change in the structure and function of chromosomes in the cell nucleus. The lncRNA XIST, for example, induces the inactivation of one of the two X chromosomes in female mammals. By interacting with mRNAs, lncRNAs can regulate the splicing of mRNAs in a similar way to short non-coding RNAs (siRNAs and miRNAs). LncRNAs also regulate various proteins that are responsible for DNA repair and the control of cell division (Statello et al., 2020).

In addition, many lncRNAs are also localised in the cytoplasm or in organelles. By binding to mRNAs, for example, they can alter their stability and translation rate or influence the transmission of signals within the cell. Other lncRNAs bind miRNAs or proteins and reduce their availability (Noh et al., 2018).

The various functions of lncRNAs in the regulation of essential cellular processes – such as development, DNA repair and cell division – are the reason why some incorrectly regulated lncRNAs can contribute to the development of cancer. These play a role in particular in the formation of metastases, immune escape, changes in cell metabolism and vascularisation in tumours. They are also important in diseases of the vascular and nervous system and in diabetes (Mercer et al., 2022).

5.2 Description of the procedure and applications

As lncRNAs can regulate cellular processes, they also have the potential to be used as active substances themselves. However, the size of the molecules makes it difficult to deliver them into the cell; furthermore, they may activate the immune system. This could be prevented by isolating the functional regions and constructing a shorter RNA with the same properties. This has already been achieved in the mouse model with the lncRNA NRON, which can slow down bone loss in osteoporosis (Mercer et al., 2022).

5.3 State of development

So far, lncRNA-based therapies have only been tested in preclinical studies. Even though various mouse studies have already shown that lncRNA-based therapies are possible in principle, no candidate has yet made it into the clinical phase (Mercer et al., 2022).

5.3.1 Challenges

The transport of long RNAs in sufficient quantities into the right tissues and cells, and additionally into the desired cell compartments, has proven to be difficult. In addition, off-target effects are relatively common. Studies in animals are complicated by the fact that lncRNAs are relatively poorly conserved evolutionarily and therefore lncRNAs in experimental animals sometimes differ greatly from their counterparts in humans. In such cases, it is often necessary to genetically modify model organisms accordingly. In preclinical cancer studies, human tissue is also transplanted into model organisms or artificial organs, so-called organoids. (Mercer et al., 2022).

5.4 Possible adverse effects

Due to their length, the risk of lncRNAs binding unintended RNA or DNA sequences is relatively high. As lncRNAs are normally only present in small quantities, there is also a risk that too high a dose could lead to the binding of proteins that are needed for other cellular tasks. In addition, the lncRNAs can enter non-target cells. It is still unclear how strongly the human immune system reacts to lncRNAs (Mercer et al., 2022).

6 RNA-directed DNA methylation

6.1 Introduction

In plants, short non-coding RNAs can regulate gene expression not only after transcription, as described in Chapter 2 'Post-transcriptional gene silencing', but also before transcription. This process is known as RNA-directed DNA methylation (RdDM). Methyl groups are added to the DNA, making genes located in this DNA region less accessible to proteins and therefore less likely to be transcribed. This process is often used naturally to suppress genes called transposons, which can otherwise jump around in the genome and influence other genes. However, it can also regulate genes that are involved in the development or reproduction of the plant. Finally, RdDM can also be a response of the plant to stress. DNA methylation can be inherited over several generations (Erdmann & Picard, 2020). How DNA is remethylated in humans and other animals is still relatively poorly understood.

6.2 Description of the procedure

As in RNA interference, a double-stranded RNA is first truncated and then bound by a protein from the Argonaute family (AGO). In contrast to post-transcriptional gene silencing, however, the RNA-AGO complex does not bind a functional mRNA, but an RNA that is currently being transcribed and is therefore in the vicinity of the complementary DNA sequence. After binding, this complex recruits various proteins that methylate DNA in this region (see Figure 9). The methylations can also lead to changes in proteins linked to the DNA (so-called histones). This modifies the structure of the chromosomes and can make the DNA in this region less accessible to proteins (Erdmann & Picard, 2020).



Figure 9: RNA-mediated DNA methylation in Arabidopsis thaliana.

a) A DNA sequence is transcribed by polymerase IV.
b) The enzyme RDR2 synthesises a second strand for the single-stranded RNA.
c) The dsRNA is processed into siRNAs by DCL3 and methylated by HEN1.
d) Together with AGO4, the siRNA binds a Pol-V transcript and recruits the protein DRM2, which methylates the DNA sequence (Kumar & Mohapatra, 2021).

6.3 Possible applications and state of development

The RdDM process can be used to suppress certain genes specifically and over several generations without changing the DNA sequence. This can be achieved, for example, by integrating a short RNA sequence specific for the target gene into a virus. The virus then introduces the short RNA sequence into the plant cells.

So far, the flower colour of petunias and the fruit ripening of tomatoes have been changed in this way. However, no application is yet ready for the market (Erdmann & Picard, 2020).

DNA methylation patterns can also be modified using the CRISPR-Cas9 system. This uses a Cas9 protein that has been modified so that it cannot cleave DNA molecules. Instead, it can be linked to proteins that methylate the DNA. By binding a specific DNA sequence, the guide RNA brings the complex into the vicinity of the DNA region that is to be methylated (Papikian et al., 2019). This system is also being tested in human cell cultures in order to suppress cancer-causing genes in cancer cells (McDonald et al., 2016). However, this technique, known as 'epigenetic editing', differs significantly from the natural RdDM described above

7 RNA activation

7.1 Introduction

In order for cells to function correctly, a large number of genes must be regulated so that the right proteins are produced in the right quantity at the right time in the right place in the cell. Many diseases can be traced back to faulty gene regulation and the resulting overproduction or underproduction of certain proteins. Correcting these incorrect regulations therefore lends itself to targeted therapies. Chapter 2 'Post-transcriptional gene silencing' presents methods that use short pieces of RNA, called small interfering RNAs (siRNAs) and antisense oligonucleotides (ASOs), to reduce or completely switch off the activity of genes. On the contrary, short RNA oligonucleotides can also increase the transcription of a gene. These processes are called RNA activation (RNAa), corresponding to RNA interference (RNAi), and the short noncoding RNAs involved are called small activating RNAs (saRNAs). This utilises a natural regulatory mechanism of the cell (Tan et al., 2021).

7.2 Description of the procedure

saRNAs are chemically synthesised, -stranded RNA oligonucleotides with a length of 21 nucleotides. Like siRNAs, saRNAs are also bound by a protein of the Argonaute family (usually AGO2) and an RNA strand is then removed (see Section 2.2.1.2 'Small interfering RNA'). The saRNA and AGO protein are imported together into the cell nucleus, where the saRNA binds to a specific DNA sequence by base pairing. This is a so-called promoter sequence. Promoter sequences are essential for regulating and initiating transcription of a specific gene. After binding to the promoter, the saRNA and the AGO protein together with other proteins form the RNA-induced transcriptional activation (RITA) complex, which promotes transcription of the gene. In addition, certain proteins that interact with the DNA, so-called histones, are often modified in such a way that the gene can be read more easily (see Figure 10). However, what exactly happens during the process of RNA activation and which proteins are involved is still only partially known (Portnoy et al., 2016).



Figure 10: RNA activation.

The saRNA is taken up into the cell as dsRNA by endocytosis and bound to AGO2. An RNA strand is then removed. After leaving the endosome, the saRNA and AGO2 bind a DNA sequence and, together with RHA, CTR9 and DDX5, form the RITA complex, which leads to an increase in transcription of a gene (Tan et al., 2021).

7.3 Applications and state of development

A number of saRNAs have already been tested in cell cultures or animal models. In most of these studies, attempts have been made to increase the expression of tumour-suppressing genes in cancer cells. This has resulted in a slowdown in tumour growth in various types of cancer. The most advanced is the development of the saRNA MTL-CEBPA, which activates the CEBPA gene that is downregulated in many cancer cells. Phase Ia and Ib clinical trials have been carried out on patients with myeloid leukaemia. Treatment with lipid nanoparticlecoated MTL-CEBPA was successful. A phase II trial is planned for 2022 (Tan et al., 2021).

In principle, saRNAs can be developed for all genes for which a promoter sequence is known. Bioinformatic methods can be used to identify the target sequence and design the saRNA. In addition to the aforementioned cancer therapies, treatments for diseases of the nervous system or metabolism caused by the dysregulation of individual genes are promising (Tan et al., 2021).

As with siRNAs, chemical modifications are also used for saRNAs in order to increase stability and reduce the immune response. Lipid nanoparticles, hybrid nanoparticles and GalNac are used for cell-specific delivery (see Section 2.2.1.2 'Small interfering RNA') (Tan et al., 2021).

7.3.1 Challenges

Since saRNAs are chemically and structurally very similar to siRNAs, similar challenges arise with regard to the stability and specificity of the molecules (see Chapter 2 'Post-transcriptional gene silencing').
8 Circular RNAs

8.1 Introduction

Circular RNAs (circRNAs) are single-stranded RNAs that are produced in the cell primarily during splicing of pre-mRNAs to mRNA (see Section 3.2 'Description of the procedure') and are relatively stable in the cell. They can code for proteins as well as be non-coding. CircRNAs differ from long non-coding RNAs (lncRNAs) and messenger RNAs (mRNAs) in that their ends are not terminated by cap and poly(A) structures but bind to each other covalently. In human cells, repetitive sequences in particular form circular structures. Although the first circular RNAs had already been discovered in the 1970s, it was only in the last ten years that researchers realised how frequently this class of RNAs occurs in the cell (C. X. Liu & Chen, 2022).

Since then, it has been discovered that non-coding circ-RNAs can interact with other RNAs as well as with DNA and proteins. As a result, they play very diverse roles in gene and cell regulation (see Figure 12). Like lncRNAs, some circRNAs can bind DNA segments and form so-called R-loops (see Chapter 5 'Long non-coding RNAs'). These loops influence the replication, repair and/or transcription of DNA. By binding to proteins, they can influence various cellular processes. The translation and splicing of mRNAs can also be regulated by circRNAs. circRNAs, which have several binding sites for certain proteins or miRNAs (see Chapter 2 'Post-transcriptional gene silencing'), are known as protein or miRNA sponges. They can reduce the concentration of these proteins or miRNAs in the cell. In addition to non-coding circRNAs, there are also protein-coding circRNAs (C. X. Liu & Chen, 2022).



Figure 11: circRNA-based medical applications.

A) RNA aptamer B) miRNA sponge C) Protein sponge D) Circular antisense RNA E) Immune response boosters (not part of the report) F) Immune response inhibitors (not part of the report) G) Translatable circRNAs H) circRNAs as targets of therapy (C. X. Liu & Chen, 2022).

8.2 Description of the procedure and applications

Because circRNAs are more stable in the cell and are less frequently recognised by the immune system than linear RNAs, they are of particular interest for medical applications. Artificial circular RNAs are being tested for various applications. The ends of RNA aptamers can be joined after synthesis, which preserves their activity for longer (see Figure 11 A). circRNAs can also be designed to contain multiple binding sites for specific miRNAs or proteins, thereby serving as molecular sponges that bind these molecules and thus reduce them in the cell (see Figure 11 B and C). Circular antisense RNAs function in a very similar way to linear ASOs and modulate the stability, structure and activity of mRNAs through binding (see Figure 11 D and Section 2.2.1.1 'Antisense oligonucleotides'). Some circRNAs can also modulate the innate immune system by activating or blocking certain receptors (see Figure 11 E and F). In addition, in vitro-transcribed, protein-coding circRNAs have the potential to serve as a more stable alternative for mRNAs (see Figure 11 G). Since some circRNAs can contribute to the development of cancer, they are also potential targets for therapies. In addition, circRNAs are over-expressed or underexpressed in certain diseases, which could allow them to serve as biomarkers for these conditions (see Figure 11 H) (C. X. Liu & Chen, 2022).

circRNAs can be produced either in vitro or in vivo. In in vitro synthesis, a linear RNA is first produced and its ends are joined together either chemically or biochemically using enzymes or ribozymes (see Chapter 9 'Ribozymes'). In the in vivo method, the genetic material is introduced into a cell on a plasmid, where the RNAs are expressed and circularised by natural cellular mechanisms (Obi & Chen, 2021).

8.3 State of development

Artificial circRNAs are a very new technology that is still in its early stages. Most applications have so far been developed and tested in cell cultures. However, since circRNAs are similar to other RNAs for which there is already better existing research (e.g. siRNAs, aptamers or mRNAs) it can be assumed that, depending on their functional design, many findings from research into these RNAs can be transferred (C. X. Liu & Chen, 2022).

9 Ribozymes

9.1 Introduction

In the 1980s, it was discovered that not only proteins but also RNAs can catalyse chemical reactions. These RNAs were named ribozymes after enzymes. The reaction spectrum of naturally occurring ribozymes is relatively narrow: they mainly catalyse the formation of proteins and the cleavage or ligation of RNA. However, the repertoire of possible binding partners and the reaction spectrum has since been significantly increased by the development of synthetic ribozymes (Balke & Müller, 2019).

9.2 Description of the procedure

Ribozymes for therapeutic applications can be produced either by in vitro selection or by rational design for a specific function. In vitro selection is performed using the SELEX process, which was developed for aptamers. This involves selecting from a pool of random pieces of RNA those that have the desired properties (see Section 4.2.3 'SELEX'). In rational design, natural ribozymes are specifically adapted to a new function. However, this requires a great deal to be known about the other molecules involved in the catalysed reaction. So far, three ribozymes derived from viruses have been used for this purpose: Hammerhead (HH), Hairpin (Hp) and Hepatitis Delta Virus (HDV). These three ribozymes can be modified in such a way that, similar to siRNAs, they cleave specific RNA sequences and thereby inactivate them. However, they can also be used to remove harmful mutations by alternative splicing of a pre-mRNA and thereby restore protein function (see Section 3.2 'Description of the procedure'). It is also possible to repair other RNAs, such as mRNAs, by exchanging mutated sequences. These applications of ribozymes that have RNAs as target molecules do not alter the genetic material. However, it is also possible in principle to use ribozymes to exchange certain DNA sequences and thus alter the genome (Balke & Müller, 2019).

9.3 Possible applications and state of development

Ribozymes have already been successfully used in cell cultures and animal models to combat viruses, including HIV and hepatitis B and C. Promising results were also obtained in the treatment of various cancer cells. However, all subsequent clinical trials were cancelled between 2002 and 2009 due to a lack of efficacy or the risk of serious side effects, at the latest in phase III. Since then, hardly any research has been carried out in this area. On the one hand, this has to do with the difficulties in the application of ribozymes (see Section 9.3.1 'Challenges'), but also with the development of RNAi and CRISPR-Cas as methods for gene regulation and gene editing respectively. These new methods are considered more effective in these areas of application and the corresponding constructs are easier to produce than ribozymes. However, ribozymes could still be suitable for certain applications such as the repair of mRNAs (Balke & Müller, 2019).

9.3.1 Challenges

As with all RNA-based therapeutics, stability, transfer into the cell and recognition by the immune system are hurdles to efficacy. As with ASOs or siRNAs, ribozymes also use 2'-O-methyl, 2'-fluoro or 2'-O-methoxyethyl modifications or artificial nucleotides such as LNAs (locked nucleic acids) to avoid enzymatic cleavage and recognition by the immune system. This can also be achieved using ribozymes from L-enantiomers of natural RNA. A further difficulty is the ion concentration inside cells, which is not ideal for the function of ribozymes and requires complex optimisation (Balke & Müller, 2019).

10 Delivery

All the RNA technologies presented are based on the fact that the resulting constructs can perform their function in certain cells – or bind to them. However, RNA molecules are very unstable and susceptible to enzymatic degradation. In addition, they cannot penetrate cell membranes due to their negative charge. In plant cells, the cell walls represent an additional barrier. The difficulty of introducing RNAs into the target cells has long prevented the successful use of these technologies outside the Petri dish. Only the development of delivery methods that enable the RNAs to be transported to the target site and penetrate the target cells has helped these technologies to achieve a breakthrough.

10.1 Medical delivery methods

For medical use, RNA-based therapeutics must fulfil a number of requirements. They must enter the targeted tissue without being absorbed by other tissues or triggering too strong an immune response; they must interact with the correct cell type in the target tissue; and they must be taken up into the target cells (Paunovska et al., 2022).

10.1.1 Naked RNA

In principle, RNA can be injected 'naked', i.e. without a coating, under the skin, into the muscle or into the cerebrospinal fluid. Small RNAs, such as siRNAs, can be chemically modified for stabilisation and combined with molecules that bind to receptors on the cell surface and thus enable cell entry. In particular, conjugation with N-acetylgalactosamine for entry into liver cells has proven successful, but other small biomolecules, lipids, antibodies or aptamers can also be used. However, siRNAs are often recognised by the immune system and degraded or excreted via the kidneys, which greatly reduces their effectiveness (see Section 2.3.3.1 'Challenges'). Naked mRNA can also be taken up into the cell via an as yet unknown mechanism. A method has also been developed in which mRNA is injected directly into dendritic cells in vitro to stimulate the production of antigens (Xiao et al. 2022).

10.1.2 Non-viral vectors

Research into alternative medical delivery methods began as early as the 1970s. These should make it possible to stabilise RNAs and other biologically active molecules – such as cancer drugs – and protect them from degradation by enzymes. Spherical lipid nanoparticles have proven to be particularly promising.

10.1.2.1 Lipid nanoparticles and liposomes

Lipid nanoparticles (LNPs) and liposomes are spherical particles that are made up of lipids (fat-like substances) and can enclose molecules. They are taken up into the cells via the process of endocytosis. In the process, they are enclosed in another double lipid layer, called an endosome, from which they must escape again in order to release the transported molecules into the cytoplasm, where the RNA becomes functional (see Figure 12). Lipid nanoparticles are currently the most important delivery method for RNA therapeutics. They have already been clinically tested and approved in various formulations for the delivery of siRNAs and mRNA vaccines.

Lipids consist of a water-attracting head group and a waterrepellent tail group and are found in all membrane systems in cells. Due to their properties, they line up in an aqueous environment with the head group facing outwards and the tail group facing inwards and form spherical structures consisting of one or two lipid layers. The lipid layers can consist of different types of lipids. The ratio between the size of the head and tail groups, the charge of the head group and the saturation of the tail group of these lipids determine which structures are formed. In nature, the small single-layer micelles and the larger double-layer liposomes occur, which form a cavity. Hydrophilic molecules can be transported in this cavity. Hydrophobic molecules can be trapped in the double lipid layer. Liposomes can have one (unilamellar vesicles) or several double lipid layers (multilamellar vesicles) and a diameter of 20 to over 1000 nanometres. The size of the liposomes has an influence on how quickly they are recognised by the immune system. Naturally occurring lipids are either negatively or neutrally charged. The type and strength of the charge depends on the head groups (Paunovska et al., 2022; Tenchov et al., 2021).

Positively charged (cationic) lipids are suitable for the delivery of RNA active substances. As RNAs are negatively charged, they interact with aqueous environments through hydrophobic and electrostatic interactions. However, positively charged lipids have been shown to be harmful to body cells. Therefore, today ionisable lipids are mainly used, which are positive in an acidic environment but become neutral under physiological conditions

in the body. This allows the mRNA to bind to the cationic lipids during production in an acidic environment. After injection, the lipids are neutral until they become positively charged again in the endosome at a lower pH, allowing the RNA to enter the cytosol. The pH in the cytosol is then higher again and the lipids are uncharged. In contrast to negatively or dipolar charged lipids, which also occur naturally, both positively charged and ionisable lipids are produced synthetically. In addition, various auxiliary substances are also used, such as cholesterol, to simplify cell entry and increase stability. The addition of the polymer PEG to lipids serves, among other things, to stabilise the lipid shell and bypass the immune system. Paradoxically, however, PEG can also trigger a strong immune reaction (see Section 3.5 'Undesirable effects'). The composition of the envelope can be optimised for different applications and forms of administration (intravenous, intramuscular, subcutaneous, etc.) and target tissues (Zak et al., 2021).

10.1.2.2 Polymer-based nanoparticles

Polymer nanoparticles are another way of introducing RNAs into cells. These consist of chains of similar units that can be chemically modified. This makes it relatively easy to change the electrical charge, stability and molecular weight of the particles. A frequently used carrier material is the biodegradable polyester polylactide-co-glycolide (PLGA), which is also used in implants and surgical sutures. As PLGA is neutral, positive chemical groups are added so that it can interact electrostatically with the negatively charged RNA. Other polymers such as polyethyleneimine (PEI), polylysine (PLL) and poly(beta-amino ester) (PBAE) are already positively charged. However, PEI and PLL are poorly tolerated and must be chemically modified before use (Paunovska et al., 2022). The poor compatibility of cationic particles can also be circumvented by using ionisable polymers such as aminopolyester (APE). Polymer nanoparticles, similar to lipid nanoparticles, are taken up into the cell by endocytosis. However, they have the advantage that they can bind larger RNA molecules and that they can more easily escape from the endosome and release the RNA into the cell interior (Gupta et al., 2021).

10.1.2.3 Cell-penetrating peptides

Cell-penetrating peptides are a group of linear and branched peptides consisting of five to 30 amino acids that can penetrate the cell membrane. What they have in common is that they have a positive charge and are both water-attracting and fat-attracting. These properties enable them to be taken up into the cell, probably via endocytosis. By being linked to an active substance either by covalent bonding or complexation, they can transport it into the cell interior. In order to improve the physical and chemical properties of the peptides, they are often modified with polymers (e.g. PEG) or fatty acids. Cell-penetrating peptides are mainly used for the delivery of siRNAs (Falato et al., 2021).

The siRNA can be covalently bound to the peptide by means of a separable compound. This compound, for example a disulphide bond, can be separated in the cytoplasm after transport through the cell membrane, whereby the siRNA is released from the peptide and can bind to an AGO protein. However, such binding is difficult to achieve in practice, and the efficiency of silencing has so far been insufficient in most experiments (Falato et al., 2021).

In contrast, the complexation of the negatively charged siRNA with the positively charged peptide is relatively simple and does not require a chemical reaction or purification. However, the rather weak binding can lead to instability and premature release of the siRNA due to electrostatic interaction. It is therefore important to find the right ratio between peptide and RNA. The structure of the peptide used is also important for the stability of the complex. This is why experiments are being carried out with different peptides. In experiments, this method showed some promising results both in vitro and in vivo (Falato et al., 2021). By using the right peptides, it has even been possible to introduce siRNAs into certain cell types of the brain (Youn et al., 2014).

Cell-penetrating peptides can also be complexed with lipids to form lipid-peptide hybrid nanoparticles. Similar to the lipid nanoparticles described above, these can enclose an siRNA and transport it into the cell interior (Falato et al., 2021).

The method shows promising research results and there have been several clinical trials with cell-penetrating peptides since 2000. However, none of the approximately 1,850 known cell-penetrating peptides have yet been approved for therapeutic purposes (Tripathi et al., 2018). This is probably due, among other things, to the often still insufficient stability in vivo, lack of specificity and immune reactions (Falato, Gestin, and Langel 2021).

10.1.3 Applications and state of development

As early as the 1970s, attempts were made to use nucleic acids packed in nanoparticles as medicines. At first, polymer nanoparticles were used for this purpose. Later it was shown that it was possible to coat RNAs with cationic liposomes. However, these particles tended to aggregate or were recognised and broken down by the immune system. It was not until around 2010 that most of the problems associated with the use of lipid nanoparticles were solved. Patisiran, the first drug with siRNAs encapsulated in lipid nanoparticles, was approved in both the USA and Europe in 2018. Another major step was taken in December 2020 with the FDA approval of Comirnaty® and Spikevax®, two mRNA vaccines against SARS-CoV-2. The vaccines were developed by Pfizer and Moderna respectively and both use lipid nanoparticles. In addition to these two authorised vaccines, a number of other SARS-CoV-2 vaccines are in development (Gupta et al., 2021). There are also mRNA vaccines being developed against various other viruses, including all influenza viruses (Arevalo et al., 2022). Several mRNA vaccines are already in the clinical phase (Lorentzen et al., 2022) as well as siRNAs, miRNA mimics and ASOs against various types of cancer (Gupta et al., 2021). Research is also being conducted into new ways of administering mRNAs, such as through nasal sprays (Xiao et al., 2022).

10.1.3.1 Challenges

For many functions, it is essential that the RNAs reach the desired tissues and cells. However, nanoparticles tend to accumulate in the liver when administered intravenously, which limits the applications of the technologies. Research is therefore being carried out into methods that make it possible to direct the nanoparticles into the desired tissues and cells. Targeting can be either passive or active. With passive targeting, the size and composition of the particles are selected so that they remain in the blood circulation for a long time without accumulating in the liver or being excreted or recognised by the immune system (e.g. through the binding of PEG to the nanoparticle surface). On the other hand, the formulation of the particles also enables them to be transported into the target tissue by the cell's own mechanisms. This is mainly achieved by certain proteins attaching to the surface of the particles. Recognition by the immune system can lead to the particles being transported to lymphatic organs such as the spleen. This removes the particles from circulation, but an immune reaction against components of the particles is still possible. Active targeting uses molecules that bind to surface proteins of the target cells. This binding enables the nanoparticles to subsequently enter the cell interior by endocytosis (Paunovska et al., 2022). For the treatment of tumour cells, for example, tumour-specific antibodies are attached to the nanoparticles (Zak et al., 2021). Crossing the blood-brain barrier is particularly difficult for RNA applications in brain cells (Xiao et al., 2022). Nusinersen is therefore injected directly into the spinal fluid of the spinal column (Claborn et al., 2018). Research is also being conducted into ways of briefly weakening the blood-brain barrier using ultrasound (Kovacs et al., 2017) or to modify ASOs so that they can diffuse through the blood-brain barrier (Relizani et al., 2017).

10.1.4 Undesirable effects

In addition to isolated severe side effects, mRNA vaccinations often also resulted in strong immune reactions, which can be attributed primarily to the lipid components (Klimek et al., 2021). These could be reduced by improving the composition of the liposomes. Among other things, attempts are being made to combine ionisable lipids with exosomes and cell membranes of human cells (Xiao et al., 2022). The choice of injection site also plays a role.

10.2 Delivery in plant protection

Plant cells are surrounded by a solid cell wall and also protected from the outside by a layer of wax. These obstacles have so far made the delivery of RNA in plants considerably more difficult and are one of the reasons why research has long been focused on transgenically expressed RNAs for plant protection. However, this approach is not dealt with in this report. The above-mentioned obstacles must be overcome when administering exogenous RNA.

RNA sprays are used to spray double-stranded RNAs onto the plant surface. If the RNA is to be absorbed by a chewing insect, it is sufficient for the RNA to remain on the leaf surface. There it must withstand UV radiation and rain. However, if stinging-sucking insects or viruses or fungi are to be reached, systemic dissemination in the plant is required. To do this, RNAs must first overcome the cuticle, a water-impermeable, waxy coating of the leaf, and enter the epidermis. Alternatively, dsRNA can also be administered via the roots or by injection (Zotti et al., 2018). In either case, the molecules must then penetrate the cell wall and the cell membrane. The RNA must remain stable and must not be degraded by nucleases. If the RNA is to be effective in an insect, it must also remain stable in the intestinal tract (Hoang et al., 2022). The formulation and method of delivery are therefore of paramount importance (Christiaens et al., 2020; K. Y. Zhu & Palli, 2020).

10.2.1 Naked RNA

Double-stranded RNA can be sprayed onto the leaves in a solution. It remains intact on the leaf surface for different lengths of time depending on the environmental conditions and the nature of the leaf. Naked RNA can enter the cell interior, especially if the RNA solution is sprayed at high pressure. However, uptake is not very effective and depends on the target plant. In order to avoid the obstacle of the cuticle, the leaves are therefore often mechanically inoculated in the laboratory before application. This means that the dsRNA solution is rubbed onto the leaves



Figure 12: Medical delivery methods.

Left: 1) A lipid nanoparticle with ionisable lipids, helper lipids, PEG, and cholesterol 2) A polymer nanoparticle 3) A liposome with aptamer 4) A spherical nucleic acid (not discussed in the report) 5) A DNA nanostructure. Right: Endocytosis. A lipid nanoparticle with RNA binds to a surface protein and is entrapped in an endosome. Due to the lower pH in the endosome, the RNA is released from the nanoparticle and escapes from the endosome. The RNA is translated by the ribosomes in the endoplasmic reticulum (Y. Zhu et al., 2022).

by hand so that the walls of superficial cells break open and absorb the RNA. However, this method is not suitable for large-scale application in the field (Rank & Koch, 2021).

10.2.2 Mini cells

Escherichia coli bacteria can be made to produce small cells that contain proteins, lipids and RNAs, but no chromosome, by switching off the so-called min genes. These minicells cannot grow or divide. If the bacterium also produces dsRNAs, these can be packaged into minicells. Enclosed in minicells, the dsRNAs are protected from degradation by nucleases (Islam et al., 2021).

10.2.3 Nanoparticles

Synthetic nanoparticles are already used for the delivery of RNAi into animal cells (see Section 10.1.2.2 'Polymer-

based nanoparticles'). However, these particles have a diameter of 100 to 200 nanometres and are therefore 10 to 20 times too large to diffuse through the cell wall. As in animal cells, the nanoparticles must be able to enter the cell by endocytosis, then leave the endosome again and release the RNA into the cell interior. Particles of various sizes and shapes have been developed for this purpose, although most of them are still in the early stages of development (Schwartz et al., 2020; Shidore et al., 2021).

10.2.3.1 Layered double hydroxide

Layered double hydroxide (LDH), also known as Bio-Clay or NanoClay, is a nanomaterial consisting of layers of a positively charged magnesium alloy. The nanolayers bind RNA and can be sprayed onto the leaves when dissolved in water. Initially, aluminium was used as the metallic component (MgAl-LDH). Because aluminium has been suspected to be toxic, it has since been replaced by iron (MgFe-LDH). This alloy is considered non-toxic and biodegradable (Jain et al., 2022). The acidic environment on the leaf surface, which occurs when the carbon dioxide released by the plant at night reacts with water from the humidity in the air to form carbonic acid, helps to break down the NanoClay and leads to the release of the dsRNA onto or into the plant. In tobacco plants, dsR-NAs bound to NanoClay showed activity against PMMo virus for at least 20 days, compared to only five days after spraying naked dsRNAs (Shidore et al., 2021).

10.2.3.2 Carbon quantum dots (CQDs)

Carbon quantum dots are carbon nanoparticles that can form a complex with RNA by attaching positive organic compounds. This protects the RNA from enzymatic degradation by nucleases and optimises uptake by the cells. A nanocomplex with siRNA has an average diameter of 3.5 nanometres and can diffuse through the cell wall. Initial trials with siRNA in Nicotiana benthamiana showed a good silencing effect. However, it turned out that small nanocomplexes that could diffuse well through the cell wall were poorly taken up by the cell and, conversely, large complexes could only diffuse poorly through the cell wall (Schwartz et al., 2020). Quantum dots synthesised from PEG are considered harmless. However, the concentration required for this application and the toxicity at this concentration still need to be evaluated (Shidore et al., 2021).

10.2.3.3 Gold nanoparticles

Gold particles with a diameter of less than 20 nanometres can penetrate the cell wall. Modified DNA strands that can bind siRNAs are attached to the surface of the particles. Whereas rod-shaped particles are taken up into the cell by endocytosis, this is not the case with spherical particles. However, both forms can release the bound siRNAs into the cell (H. Zhang et al., 2021).

10.2.3.4 DNA nanostructures

In nature, DNA molecules form double helix structures. However, it is possible to design DNA strands in such a way that they form other structures – for example a tetrahedron. In addition, binding sites for RNA, DNA or proteins can be incorporated. It has been shown that siRNAs bound to such DNA nanostructures can be taken up by cells. The efficiency of uptake depends on the size and mechanical stiffness of the structure. The smaller and stiffer it is, the easier it is taken up. Surprisingly, the siR-NAs trigger different silencing processes depending on the structure to which they were bound (H. Zhang et al. 2019).

10.2.3.5 Chitosan nanoparticles (CNPs)

Chitosan is the deacetylated form of the biopolymer chitin, which is found in nature primarily in the exoskeleton of insects and in some fungi. The positively charged chitosan forms a very stable complex with RNA, which protects it from enzymatic degradation. CNPs are biodegradable and non-toxic (Shidore et al., 2021). They have also been shown to protect against the acidic environment in the gut of insects (Kolge et al., 2021).

10.2.3.6 Single-walled carbon nanotubes (SWCNT)

SWCNTs are nanotubes made of carbon, which are produced by rolling up a graphene layer at a specific angle. Due to their small diameter of less than 20 nanometres, they can enter plant cells through the cell wall. siRNAs can be absorbed by the SWCNTs by detaching the sense strand from the antisense strand and binding the two strands separately to the SWCNT. In the cell, the strands detach from the nanotube and form a double-stranded siRNA. Concerns about toxicity in mammals limit the applicability of SWCNTs, especially in edible plants (Shidore et al., 2021).

10.2.4 State of development

Nanoparticles or microcells have the potential to significantly increase the stability and specificity of exogenous RNA and thus lower two of the biggest hurdles for the successful application of RNAi in crop protection. So far, however, most of the applications of the technologies described above have only been proof-of-concept experiments that show that the methods work in principle. These experiments have been carried out in isolation and on a small scale. However, there is little data on how these preparations behave under real conditions and how long they remain intact in nature (Rank & Koch, 2021).

So far, little is known about why RNA sprays are effective on some plants and not on others. There is also a lack of knowledge about how the various nanoparticles are absorbed into the plant and what unintended effects they may trigger (Rank & Koch, 2021).

References

Abdeeva IA, Maloshenok LG, Pogorelko GV, Mokrykova MV, Bruskin SA (2019) **RNA-aptamers—As targeted inhibitors of protein functions in plants.** Journal of Plant Physiology, 232, 127–129. https://doi.org/10.1016/J.JPLPH.2018.10.026

Abdellatef E, Will T, Koch A, Imani J, Vilcinskas A, Kogel KH (2015) Silencing the expression of the salivary sheath protein causes transgenerational feeding suppression in the aphid Sitobion avenae. Plant Biotechnology Journal, 13 (6), 849–857. https://doi.org/10.1111/PBI.12322

Adachi H, Hengesbach M, Yu YT, Morais P (2021) From antisense RNA to RNA modification: Therapeutic potential of RNA-based technologies. In Biomedicines (Vol. 9, Issue 5). MDPI AG. https://doi.org/10.3390/biomedicines9050550

Aida V, Pliasas VC, Neasham PJ, North JF, McWhorter KL, Glover SR, Kyriakis CS (2021) Novel Vaccine Technologies in Veterinary Medicine: A Herald to Human Medicine Vaccines. Frontiers in Veterinary Science, 8, 340. https://doi.org/10.3389/fvets.2021.654289

Arevalo CP, Bolton MJ, Le Sage V, Ye N, Furey C, Muramatsu H, Alameh M-G, Pardi N, Drapeau EM, Parkhouse K, Garretson T, Morris JS, Moncla LH, Tam YK, Fan SHY, Lakdawala SS, Weissman D, Hensley SE (2022) A multivalent nucleoside-modified mRNA vaccine against all known influenza virus subtypes. Science, 378 (6622), 899–904. https://doi.org/10.1126/science.abm0271

Bachman P, Fischer J, Song Z, Urbanczyk-Wochniak E, Watson G (2020) Environmental Fate and Dissipation of Applied dsRNA in Soil, Aquatic Systems, and Plants. Frontiers in Plant Science, 11, 508351. https://doi.org/10.3389/FPLS.2020.00021

Bachman PM, Bolognesi R, Moar WJ, Mueller GM, Paradise MS, Ramaseshadri P, Tan J, Uffman JP, Warren J, Wiggins BE, Levine SL (2013) **Characterization of the spectrum of insecticidal activity of a double-stranded RNA with targeted activity against Western Corn Rootworm (Diabrotica virgifera virgifera LeConte).** Transgenic Research, 22 (6), 1207–1222. https://doi.org/10.1007/s11248-013-9716-5

Bachman PM, Huizinga KM, Jensen PD, Mueller G, Tan J, Uffman JP, Levine SL (2016) Ecological risk assessment for DvSnf7 RNA: A plantincorporated protectant with targeted activity against western corn rootworm. Regulatory Toxicology and Pharmacology, 81, 77–88. https://doi.org/10.1016/J.YRTPH.2016.08.001

Balke D, Müller S (2019) **Therapeutic Potential of Ribozymes.** In Agrawal S, Gait MJ (Eds.), Advances in Nucleic Acid Therapeutics (pp. 434–452). The Royal Society of Chemistry. https://doi. org/10.1039/9781788015714-00434

Barbier AJ, Jiang AY, Zhang P, Wooster R, Anderson DG (2022) **The clinical progress of mRNA vaccines and immunotherapies.** Nature Biotechnology 2022 40:6, 40 (6), 840–854. https://doi.org/10.1038/s41587-022-01294-2

Bhat B, Karve S, Anderson DG (2021) mRNA therapeutics: beyond vaccine applications. Trends in Molecular Medicine, 27 (9), 923–924. https://doi.org/10.1016/J.MOLMED.2021.05.004

Byun J (2021) Recent progress and opportunities for nucleic acid aptamers. Life, 11 (3), 1–18. https://doi.org/10.3390/LIFE11030193

Capaldi D, Teasdale A, Henry S, Akhtar N, den Besten C, Gao-Sheridan S, Kretschmer M, Sharpe N, Andrews B, Burm B, Foy J (2017) Impurities in Oligonucleotide Drug Substances and Drug Products. Nucleic Acid Therapeutics, 27 (6), 309–322. https://doi.org/10.1089/nat.2017.0691 Carvalho T (2023) Personalized anti-cancer vaccine combining mRNA and immunotherapy tested in melanoma trial. Nature Medicine. https://doi.org/10.1038/D41591-023-00072-0

Chandola C, Neerathilingam M (2019) Aptamers for Targeted Delivery: Current Challenges and Future Opportunities. Role of Novel Drug Delivery Vehicles in Nanobiomedicine. https://doi.org/10.5772/INTECHOPEN.84217

Chaudhary N, Weissman D, Whitehead KA (2021) **mRNA vaccines for infectious diseases: principles, delivery and clinical translation.** Nature Reviews Drug Discovery 2021 20:11, 20 (11), 817–838. https://doi.org/10.1038/s41573-021-00283-5

Chen L, Rashid F, Shah A, Awan HM, Wu M, Liu A, Wang J, Zhu T, Luo Z, Shan G (2015) **The isolation of an RNA aptamer targeting to p53 protein with single amino acid mutation.** Proceedings of the National Academy of Sciences of the United States of America, 112 (32), 10002–10007. https://doi.org/10.1073/PNAS.1502159112

Christiaens O, Tardajos MG, Martinez Reyna ZL, Dash M, Dubruel P, Smagghe G (2018) Increased RNAi Efficacy in Spodoptera exigua via the Formulation of dsRNA With Guanylated Polymers. Frontiers in Physiology, 9(APR), 342048. https://doi.org/10.3389/fphys.2018.00316

Christiaens O, Whyard S, Vélez AM, Smagghe G (2020) **Double-Stranded RNA Technology to Control Insect Pests: Current Status and Challenges.** In Frontiers in Plant Science (Vol. 11). Frontiers Media S.A. https://doi.org/10.3389/fpls.2020.00451

Claborn MK, Stevens DL, Walker CK, Gildon BL (2018) Nusinersen: A Treatment for Spinal Muscular Atrophy. https://doi.org/10.1177/1060028018789956, 53 (1), 61–69. https://doi.org/10.1177/1060028018789956

Crooke ST, Baker BF, Crooke RM, Liang X hai (2021) Antisense technology: an overview and prospectus. Nature Reviews Drug Discovery 2021 20:6, 20 (6), 427–453. https://doi.org/10.1038/s41573-021-00162-z

de Schutter K, Taning CNT, van Daele L, van Damme EJM, Dubruel P, Smagghe G (2022) RNAi-Based Biocontrol Products: Market Status, Regulatory Aspects, and Risk Assessment. Frontiers in Insect Science, 0, 22. https://doi.org/10.3389/FINSC.2021.818037

Dietz-Pfeilstetter A (2021) Relevance of rna interference (Rnai) as a new mode of action for plant protection with high specificity. In Journal fur Kulturpflanzen (Vol. 73, Issues 1–2, pp. 1–8). Verlag Eugen Ulmer. https://doi.org/10.5073/JfK.2021.01-02.02

EPA (2021) Pesticide Product Registration; Receipt of Application from GreenLight Biosciences to Register Ledprona as a New Active Ingredient (94614-R; 94614-E).

https://www.regulations.gov/docket/EPA-HQ-OPP-2021-0271

Erdmann RM, Picard CL (2020) **RNA-directed DNA Methylation.** PLoS Genetics, 16 (10). https://doi.org/10.1371/JOURNAL.PGEN.1009034

EPA (2023) Registration Decision for the New Active Ingredient. Ledprona (Leptinotarsa decemlineata-specific recombinant double-stranded interfering Oligonucleotide GS2). (CAS Number: 2433753-68-3). https://www.regulations.gov/document/EPA-HQ-OPP-2021-0271-0196

Escobar MA, Civerolo EL, Summerfelt KR, Dandekar AM (2001) RNAi-mediated oncogene silencing confers resistance to crown gall tumorigenesis. Proceedings of the National Academy of Sciences of the United States of America, 98 (23), 13437–13442. https://doi.org/10.1073/PNAS.241276898 Esposito R, Polidori T, Meise DF, Pulido-Quetglas C, Chouvardas P, Forster S, Schaerer P, Kobel A, Schlatter J, Kerkhof E, Roemmele M, Rice ES, Zhu L, Lanzós A, Guillen-Ramirez HA, Basile G, Carrozzo I, Vancura A, Ullrich S, ... Johnson R (2022) Multi-hallmark long noncoding RNA maps reveal non-small cell lung cancer vulnerabilities. Cell Genomics, 2 (9), 100171. https://doi.org/10.1016/j. xgen.2022.100171

Falato L, Gestin M, Langel Ü (2021) **Cell-Penetrating Peptides Delivering siRNAs: An Overview.** In Methods in Molecular Biology (Vol. 2282, pp. 329–352). Humana Press Inc. https://doi.org/10.1007/978-1-0716-1298-9_18

Feng R, Patil S, Zhao X, Miao Z, Qian A (2021) **RNA Therapeutics – Research and Clinical Advancements.** Frontiers in Molecular Biosciences, 8, 913. https://doi.org/10.3389/fmolb.2021.710738

Frankish A, Diekhans M, Jungreis I, Lagarde J, Loveland JE, Mudge JM, Sisu C, Wright JC, Armstrong J, Barnes I, Berry A, Bignell A, Boix C, Sala SC, Cunningham F, Domenico T di, Donaldson S, Fiddes IT, Girón CG, ... Flicek P (2021) **GENCODE 2021.** Nucleic Acids Research, 49 (D1), D916–D923. https://doi.org/10.1093/NAR/GKAA1087

11(1p3.//doi.org/10.1093/1010/GRAM100/

Grand View Research (2023) Global Aptamers Market Size, Share & Growth Report. https://www.grandviewresearch.com/industry-analysis/aptamers-industry

Gupta A, Andresen JL, Manan RS, Langer R (2021) Nucleic acid delivery for therapeutic applications. Advanced Drug Delivery Reviews, 178, 113834. https://doi.org/10.1016/J.ADDR.2021.113834

Hall J, Ashkinadze A, Becker JP, Laski A, Manz EM, Moravčík Š, Sjöström S, Vincent M (2022) Medicinal Chemistry of Oligonucleotide Drugs – Milestones of the Past and Visions for the Future. CHIMIA, 76 (5), 466. https://doi.org/10.2533/chimia.2022.466

Hoang BTL, Fletcher SJ, Brosnan CA, Ghodke AB, Manzie N, Mitter N (2022) **RNAi as a Foliar Spray: Efficiency and Challenges to Field Applications.** International Journal of Molecular Sciences, 23 (12). https://doi.org/10.3390/IJMS23126639

Höfle L, Biedenkopf D, Werner BT, Shrestha A, Jelonek L, Koch A (2020) **Study on the efficiency of dsRNAs with increasing length in RNA-based silencing of the Fusarium CYP51 genes.** RNA Biology, 17 (4), 463–473. https://doi.org/10.1080/15476286.2019.1700033

Hu B, Zhong L, Weng Y, Peng L, Huang Y, Zhao Y, Liang XJ (2020) Therapeutic siRNA: state of the art. Signal Transduction and Targeted Therapy 2020 5:1, 5(1), 1–25. https://doi.org/10.1038/s41392-020-0207-x

Huang ZY, Bian G, Xi Z, Xie X (2019) Genes important for survival or

reproduction in Varroa destructor identified by RNAi. Insect Science, 26 (1), 68–75. https://doi.org/10.1111/1744-7917.12513

Huvenne H, Smagghe G (2010) **Mechanisms of dsRNA uptake in insects and potential of RNAi for pest control: A review.** Journal of Insect Physiology, 56 (3), 227–235. https://doi.org/10.1016/J.JINSPHYS.2009.10.004

Islam MT, Davis Z, Chen L, Englaender J, Zomorodi S, Frank J, Bartlett K, Somers E, Carballo SM, Kester M, Shakeel A, Pourtaheri P, Sherif SM (2021) Minicell-based fungal RNAi delivery for sustainable crop protection. Microbial Biotechnology, 14(4), 1847–1856. https://doi.org/10.1111/1751-7915.13699

Jain RG, Fletcher SJ, Manzie N, Robinson KE, Li P, Lu E, Brosnan CA, Xu ZP, Mitter N (2022) Foliar application of clay-delivered RNA interference for whitefly control. Nature Plants 2022 8:5, 8 (5), 535–548. https://doi.org/10.1038/s41477-022-01152-8

Jason TLH, Koropatnick J, Berg RW (2004) **Toxicology of antisense therapeutics.** Toxicology and Applied Pharmacology, 201 (1), 66–83. https://doi.org/10.1016/J.TAAP.2004.04.017

Kang W, Bang-Berthelsen CH, Holm A, Houben AJS, Müller AH, Thymann T, Pociot F, Estivill X, Friedländer MR (2017) **Survey of 800+ data sets from human tissue and body fluid reveals xenomiRs are likely artifacts.** RNA, 23 (4), 433–445.

https://doi.org/10.1261/RNA.059725.116

Khajuria C, Ivashuta S, Wiggins E, Flagel L, Moar W, Pleau M, Miller K, Zhang Y, Ramaseshadri P, Jiang C, Hodge T, Jensen P, Chen M, Gowda A, McNulty B, Vazquez C, Bolognesi R, Haas J, Head G, Clark T (2018) **Development and characterization of the first dsRNA-**

resistant insect population from western corn rootworm, Diabrotica virgifera virgifera LeConte. PLOS ONE, 13 (5), e0197059. https://doi.org/10.1371/JOURNAL.PONE.0197059

Kim TH, Lee SW (2021) Aptamers for Anti-Viral Therapeutics and Diagnostics. International Journal of Molecular Sciences, 22 (8). https://doi.org/10.3390/IJMS22084168

Klimek L, Eckrich J, Hagemann J, Casper I, Huppertz J (2021) Allergische Reaktionen auf COVID-19-Impfstoffe – Evidenz und praxisorientiertes Vorgehen. Der Internist, 62 (3), 326–332. https://doi.org/10.1007/s00108-021-00959-5

Koch A, Biedenkopf D, Furch A, Weber L, Rossbach O, Abdellatef E, Linicus L, Johannsmeier J, Jelonek L, Goesmann A, Cardoza V, McMillan J, Mentzel T, Kogel KH (2016) An RNAi-Based Control of Fusarium graminearum Infections Through Spraying of Long dsRNAs Involves a Plant Passage and Is Controlled by the Fungal Silencing Machinery. PLOS Pathogens, 12 (10), e1005901.

https://doi.org/10.1371/JOURNAL.PPAT.1005901

Koch A, Höfle L, Werner BT, Imani J, Schmidt A, Jelonek L, Kogel KH (2019) SIGS vs HIGS: a study on the efficacy of two dsRNA delivery strategies to silence Fusarium FgCYP51 genes in infected host and non-host plants. Molecular Plant Pathology, 20 (12), 1636. https://doi.org/10.1111/MPP.12866

Koch A, Petschenka G (2022) **Exogene Anwendung von RNA zur umweltfreundlichen Bekämpfung von Schadinsekten.** Journal Für Kulturpflanzen, 74 (03–04), 75–84. https://doi.org/10.5073/JFK.2022.03-04.05

Koch A, Wassenegger M (2021) **Host-induced gene silencing – mechanisms and applications**. New Phytologist, 231 (1), 54–59. https://doi.org/10.1111/NPH.17364

Kohlberger M, Gadermaier G (2021) **SELEX: Critical factors and optimization strategies for successful aptamer selection.** Biotechnology and Applied Biochemistry. https://doi.org/10.1002/BAB.2244

Kolge H, Kadam K, Galande S, Lanjekar V, Ghormade V (2021) New Frontiers in Pest Control: Chitosan Nanoparticles-Shielded dsRNA as an Effective Topical RNAi Spray for Gram Podborer Biocontrol. ACS Applied Bio Materials, 4 (6), 5145–5157. https://doi.org/10.1021/ACSABM.1C00349

Kovacevic KD, Gilbert JC, Jilma B (2018) **Pharmacokinetics, pharmacodynamics and safety of aptamers.** Advanced Drug Delivery Reviews, 134, 36–50. https://doi.org/10.1016/j.addr.2018.10.008

Kovacs ZI, Kim S, Jikaria N, Qureshi F, Milo B, Lewis BK, Bresler M, Burks SR, Frank JA (2017) **Disrupting the blood-brain barrier by focused ultrasound induces sterile inflammation**. Proceedings of the National Academy of Sciences, 114 (1), E75–E84. https://doi.org/10.1073/pnas.1614777114

Krammer F (2020) **SARS-CoV-2 vaccines in development.** Nature 2020 586:7830, 586 (7830), 516–527. https://doi.org/10.1038/s41586-020-2798-3

Kumar S, Mohapatra T (2021) Dynamics of DNA Methylation and Its Functions in Plant Growth and Development. Frontiers in Plant Science, 12, 858. https://doi.org/10.3389/fpls.2021.596236 Li J, Liu Q, Liu J, Wu X, Lei Y, Li S, Zhao D, Li Z, Luo L, Peng S, Ou Y, Yang H, Jin J, Li Y, Peng Y (2022) An mRNA-based rabies vaccine induces strong protective immune responses in mice and dogs. Virology Journal, 19 (1), 1–13. https://doi.org/10.1186/S12985-022-01919-7/

Liu CX, Chen LL (2022) Circular RNAs: Characterization, cellular roles, and applications. Cell, 185 (12), 2016–2034. https://doi.org/10.1016/J.CELL.2022.04.021

Liu S, Jaouannet M, Dempsey DA, Imani J, Coustau C, Kogel K-H (2020) **RNA-based technologies for insect control in plant production.** Biotechnology Advances, 39, 107463. https://doi.org/10.1016/j.biotechadv.2019.107463

Liu SJ, Horlbeck MA, Cho SW, Birk HS, Malatesta M, He D, Attenello FJ, Villalta JE, Cho MY, Chen Y, Mandegar MA, Olvera MP, Gilbert LA, Conklin BR, Chang HY, Weissman JS, Lim DA (2017) CRISPRi-based genome-scale identification of functional long non-coding RNA loci in human cells. Science (New York, N.Y.), 355 (6320). https://doi. org/10.1126/SCIENCE.AAH7111

Lorentzen CL, Haanen JB, Met Ö, Svane IM (2022) **Clinical advances and ongoing trials on mRNA vaccines for cancer treatment.** The Lancet Oncology, 23 (10), e450–e458.

https://doi.org/10.1016/S1470-2045(22)00372-2

Lu RM, Hwang YC, Liu IJ, Lee CC, Tsai HZ, Li HJ, Wu HC (2020) Development of therapeutic antibodies for the treatment of diseases. Journal of Biomedical Science 2020 27:1, 27 (1), 1–30. https://doi.org/10.1186/S12929-019-0592-Z

Mahmoudi E, Soleimani R (2019) **Host-induced gene silencing of Pectobacterium carotovorum quorum sensing gene enhances soft rot disease resistance in potato plants.** Archives of Phytopathology and Plant Protection, 52 (3–4), 371–384. https://doi.org/10.1080/03235408.2019.1623503

Intps://doi.org/10.1080/03235408.2019.1623503

McDonald JI, Celik H, Rois LE, Fishberger G, Fowler T, Rees R, Kramer A, Martens A, Edwards JR, Challen GA (2016) **Reprogrammable CRISPR/Cas9-based system for inducing site-specific DNA methylation.** Biology Open, 5 (6), 866–874. https://doi.org/10.1242/bio.019067

Mercer TR, Munro T, Mattick JS (2022) **The potential of long noncoding RNA therapies.** Trends in Pharmacological Sciences, 43 (4), 269–280. https://doi.org/10.1016/J.TIPS.2022.01.008

Mills DR, Peterson RL, Spiegelman S (1967) **An extracellular Darwinian experiment with a self-duplicating nucleic acid molecule.** Proceedings of the National Academy of Sciences of the United States of America, 58 (1), 217. https://doi.org/10.1073/PNAS.58.1.217

Mishra S, Dee J, Moar W, Dufner-Beattie J, Baum J, Dias NP, Alyokhin A, Buzza A, Rondon SI, Clough M, Menasha S, Groves R, Clements J, Ostlie K, Felton G, Waters T, Snyder WE, Jurat-Fuentes JL (2021) Selection for high levels of resistance to double-stranded RNA (dsRNA) in Colorado potato beetle (Leptinotarsa decemlineata Say) using non-transgenic foliar delivery. Scientific Reports 2021 11:1, 11 (1), 1–12. https://doi.org/10.1038/s41598-021-85876-1

Mitter N, Worrall EA, Robinson KE, Li P, Jain RG, Taochy C, Fletcher SJ, Carroll BJ, Lu GQ, Xu ZP (2017) **Clay nanosheets for topical delivery of RNAi for sustained protection against plant viruses.** Nature Plants, 3. https://doi.org/10.1038/NPLANTS.2016.207

Mollocana-Lara EC, Ni M, Agathos SN, Gonzales-Zubiate FA (2021) **The infinite possibilities of RNA therapeutics.** Journal of Industrial Microbiology and Biotechnology, 48 (9–10). https://doi.org/10.1093/jimb/kuab063

Munzar JD, Ng A, Juncker D (2019) **Duplexed aptamers: history, design, theory, and application to biosensing.** Chemical Society Reviews, 48 (5), 1390–1419. https://doi.org/10.1039/C8CS00880A

Noh JH, Kim KM, McClusky WG, Abdelmohsen K, Gorospe M (2018) **Cytoplasmic functions of IncRNAs.** Wiley Interdisciplinary Reviews. RNA, 9 (3), e1471. https://doi.org/10.1002/WRNA.1471 Nudler E, Mironov AS (2004) **The riboswitch control of bacterial metabolism.** Trends in Biochemical Sciences, 29 (1), 11–17. https://doi.org/10.1016/j.tibs.2003.11.004

Obi P, Chen YG (2021) **The design and synthesis of circular RNAs.** Methods (San Diego, Calif.), 196, 85. https://doi.org/10.1016/J.YMETH.2021.02.020

OECD (2020) Considerations for the Environmental Risk Assessment of the Application of Sprayed or Externally Applied ds-RNA-Based Pesticides. Series on Pesticides, 104.

Ospina-Villa JD, Cisneros-Sarabia A, Sánchez-Jiménez MM, Marchat LA (2020) Current advances in the development of diagnostic tests based on aptamers in parasitology: A systematic review. Pharmaceutics, 12 (11), 1–22. https://doi.org/10.3390/PHARMA-CEUTICS12111046

Pallis S, Alyokhin A, Manley B, Rodrigues TB, Buzza A, Barnes E, Narva K (2022) **Toxicity of a novel dsRNA-based insecticide to the Colorado potato beetle in laboratory and field trials.** Pest Management Science, 78 (9), 3836–3848. https://doi.org/10.1002/PS.6835

Papikian A, Liu W, Gallego-Bartolomé J, Jacobsen SE (2019) Site-specific manipulation of Arabidopsis loci using CRISPR-Cas9 SunTag systems. Nature Communications 2019 10:1, 10 (1), 1–11. https://doi.org/10.1038/s41467-019-08736-7

Pardi N, Hogan MJ, Porter FW, Weissman D (2018) **mRNA vaccines** – **a new era in vaccinology.** Nature Reviews Drug Discovery 2018 17:4, 17 (4), 261–279. https://doi.org/10.1038/nrd.2017.243

Pascolo S (2021) Impfstoffe gegen COVID-19 – Aerzteverlag medinfo AG. Der Informierte @rzt.

https://www.medinfo-verlag.ch/impfstoffe-gegen-covid-19

Paunovska K, Loughrey D, Dahlman JE (2022) **Drug delivery systems for RNA therapeutics.** Nature Reviews Genetics 2022 23:5, 23 (5), 265–280. https://doi.org/10.1038/s41576-021-00439-4

Portnoy V, Lin SHS, Li KH, Burlingame A, Hu ZH, Li H, Li LC (2016) saRNA-guided Ago2 targets the RITA complex to promoters to stimulate transcription. Cell Research 2016 26:3, 26 (3), 320–335. https://doi.org/10.1038/cr.2016.22

Povsic TJ, Lawrence MG, Lincoff AM, Mehran R, Rusconi CP, Zelenkofske SL, Huang Z, Sailstad J, Armstrong PW, Steg PG, Bode C, Becker RC, Alexander JH, Adkinson NF, Levinson AI (2016) **Pre-existing anti-PEG antibodies are associated with severe immediate allergic reactions to pegnivacogin, a PEGylated aptamer.** Journal of Allergy and Clinical Immunology, 138 (6), 1712–1715. https://doi.org/10.1016/j.jaci.2016.04.058

Qin S, Tang X, Chen Y, Chen K, Fan N, Xiao W, Zheng Q, Li G, Teng Y, Wu M, Song X (2022) **mRNA-based therapeutics: powerful and versatile tools to combat diseases.** Signal Transduction and Targeted Therapy 2022 7:1, 7 (1), 1–35. https://doi.org/10.1038/s41392-022-01007-w

Rank AP, Koch A (2021) Lab-to-Field Transition of RNA Spray Applications – How Far Are We? Frontiers in Plant Science, 12. https://doi.org/10.3389/fpls.2021.755203

Relizani K, Griffith G, Echevarría L, Zarrouki F, Facchinetti P, Vaillend C, Leumann C, Garcia L, Goyenvalle A (2017) Efficacy and Safety Profile of Tricyclo-DNA Antisense Oligonucleotides in Duchenne Muscular Dystrophy Mouse Model. Molecular Therapy - Nucleic Acids, 8, 144–157. https://doi.org/10.1016/j.omtn.2017.06.013

Romeis J, Widmer F (2020) Assessing the Risks of Topically Applied dsRNA-Based Products to Non-target Arthropods. Frontiers in Plant Science, 11, 679. https://doi.org/10.3389/FPLS.2020.00679

Saad M, Faucher SP (2021) Aptamers and Aptamer-Coupled Biosensors to Detect Water-Borne Pathogens. Frontiers in Microbiology, 12, 304. https://doi.org/10.3389/fmicb.2021.643797

Schlake T, Thess A, Thran M, Jordan I (2019) **mRNA as novel technology for passive immunotherapy.** Cell. Mol. Life Sci., 76 (2), 301–328. https://doi.org/10.1007/s00018-018-2935-4

Schwartz SH, Hendrix B, Hoffer P, Sanders RA, Zheng W (2020) Carbon Dots for Efficient Small Interfering RNA Delivery and Gene Silencing in Plants. Plant Physiology, 184 (2), 647-657. https://doi.org/10.1104/PP.20.00733

Scott JG, Michel K, Bartholomay LC, Siegfried BD, Hunter WB, Smagghe G, Zhu KY, Douglas AE (2013) Towards the elements of successful insect RNAi. Journal of Insect Physiology, 59 (12), 1212-1221. https://doi.org/10.1016/J.JINSPHYS.2013.08.014

Sharun K, Tiwari R, Saied ARA, Dhama K (2021) SARS-CoV-2 vaccine for domestic and captive animals: An effort to counter COVID-19 pandemic at the human-animal interface. Vaccine, 39 (49), 7119. https://doi.org/10.1016/J.VACCINE.2021.10.053

Shen W, De Hoyos CL, Migawa MT, Vickers TA, Sun H, Low A, Bell TA, Rahdar M, Mukhopadhyay S, Hart CE, Bell M, Riney S, Murray SF, Greenlee S, Crooke RM, Liang X hai, Seth PP, Crooke ST (2019) Chemical modification of PS-ASO therapeutics reduces cellular protein-binding and improves the therapeutic index. Nature Biotechnology 2019 37:6, 37 (6), 640-650. https://doi.org/10.1038/s41587-019-0106-2

Shidore T, Zuverza-Mena N, White JC, da Silva W (2021) Nanoenabled Delivery of RNA Molecules for Prolonged Antiviral Protection in Crop Plants: A Review. ACS Applied Nano Materials, 4 (12), 12891-12904. https://doi.org/10.1021/acsanm.1c03512

Sola M, Menon AP, Moreno B, Meraviglia-Crivelli D, Soldevilla MM, Cartón-García F, Pastor F (2020) Aptamers Against Live Targets: Is In Vivo SELEX Finally Coming to the Edge? Molecular Therapy - Nucleic Acids, 21, 192-204. https://doi.org/10.1016/J.OMTN.2020.05.025

Statello L, Guo CJ, Chen LL, Huarte M (2020) Gene regulation by long non-coding RNAs and its biological functions. Nature Reviews Molecular Cell Biology 2020 22:2, 22 (2), 96-118. https://doi.org/10.1038/s41580-020-00315-9

Sun M, Liu S, Song T, Chen F, Zhang J, Huang J, Wan S, Lu Y, Chen H, Tan W, Song Y, Yang C (2021) Spherical Neutralizing Aptamer Inhibits SARS-CoV-2 Infection and Suppresses Mutational Escape. Journal of the American Chemical Society, 143 (51), 21541-21548. https://doi.org/10.1021/jacs.1c08226

Sun M, Wu Z, Zhang J, Chen M, Lu Y, Yang C, Song Y (2022) Spherical neutralizing aptamer suppresses SARS-CoV-2 Omicron escape. Nano Today, 44, 101499. https://doi.org/10.1016/J.NANTOD.2022.101499

Svoboda P (2020) Key Mechanistic Principles and Considerations Concerning RNA Interference. Frontiers in Plant Science, 11, 1237. https://doi.org/10.3389/fpls.2020.01237

Tan CP, Sinigaglia L, Gomez V, Nicholls J, Habib NA (2021) RNA Activation—A Novel Approach to Therapeutically Upregulate Gene Transcription. Molecules 2021, Vol. 26, Page 6530, 26 (21), 6530. https://doi.org/10.3390/MOLECULES26216530

Tenchov R, Bird R, Curtze AE, Zhou Q (2021) Lipid Nanoparticles from Liposomes to mRNA Vaccine Delivery, a Landscape of Research Diversity and Advancement. ACS Nano, 15 (11), 16982-17015. https://doi.org/10.1021/ACSNANO.1C04996

Traber GM, Yu A-M (2022) RNAi Based Therapeutics and Novel RNA Bioengineering Technologies. Journal of Pharmacology and Experimental Therapeutics, JPET-MR-2022-001234. https://doi.org/10.1124/JPET.122.001234

Tripathi PP, Arami H, Banga I, Gupta J, Gandhi S, Tripathi PP, Arami H, Banga I, Gupta J, Gandhi S (2018) Cell penetrating peptides in preclinical and clinical cancer diagnosis and therapy. Oncotarget, 9 (98), 37252-37267. https://doi.org/10.18632/ONCOTARGET.26442

Trougakos IP, Terpos E, Alexopoulos H, Politou M, Paraskevis D, Scorilas A, Kastritis E, Andreakos E, Dimopoulos MA (2022) Adverse effects of COVID-19 mRNA vaccines: the spike hypothesis. $\operatorname{Trends}\,\operatorname{in}$ Molecular Medicine, 28 (7), 542-554.

Tuerk C, Gold L (1990) Systematic Evolution of Ligands by Exponential Enrichment: RNA Ligands to Bacteriophage T4 DNA Polymerase. Science, 249 (4968), 505-510. https://doi.org/10.1126/SCIENCE.2200121

Vater A, Klussmann S (2015) Turning mirror-image oligonucleotides into drugs: the evolution of Spiegelmer(®) therapeutics. Drug Discovery Today, 20 (1), 147-155. https://doi.org/10.1016/J.DRUDIS.2014.09.004

vfa (2023) mRNA-Impfstoffe für Schutzimpfungen.

https://www.vfa.de/de/arzneimittel-forschung/coronavirus/ rna-basierte-impfstoffe-in-entwicklung-und-versorgung

Wang Z, Li Y, Zhang B, Gao X, Shi M, Zhang S, Zhong S, Zheng Y, Liu X, Wang Z, Li Y, Zhang B, Gao X, Shi M, Zhang S, Zhong S, Zheng Y, Liu X (2023) Functionalized Carbon Dot-Delivered RNA Nano Fungicides as Superior Tools to Control Phytophthora Pathogens through Plant RdRP1 Mediated Spray-Induced Gene Silencing. Advanced Functional Materials, 33 (22), 2213143.

https://doi.org/10.1002/ADFM.202213143

Weinstock GM, Robinson GE, Gibbs RA, Worley KC, Evans JD, Maleszka R, Robertson HM, Weaver DB, Beye M, Bork P, Elsik CG, Hartfelder K, Hunt GJ, Zdobnov EM, Amdam G V., Bitondi MMG, Collins AM, Cristino AS, Lattorff HMG, ... Wright R (2006) Insights into social insects from the genome of the honeybee Apis mellifera. Nature, 443 (7114), 931-949. https://doi.org/10.1038/NATURE05260

Werner BT, Gaffar FY, Schuemann J, Biedenkopf D, Koch AM (2020) RNA-Spray-Mediated Silencing of Fusarium graminearum AGO and DCL Genes Improve Barley Disease Resistance. Frontiers in Plant Science, 11, 476. https://doi.org/10.3389/fpls.2020.00476

Xiao Y, Tang Z, Huang X, Chen W, Zhou J, Liu H, Liu C, Kong N, Tao W (2022) Emerging mRNA technologies: delivery strategies and biomedical applications. Chemical Society Reviews, 51 (10), 3828-3845. https://doi.org/10.1039/D1CS00617G

 $Yan \; AC, \, Levy \; M$ (2018) Aptamer-Mediated Delivery and Cell-Targeting Aptamers: Room for Improvement. Nucleic Acid Therapeutics, 28 (3), 194. https://doi.org/10.1089/NAT.2018.0732

Youn P, Chen Y, Furgeson DY (2014) A Myristoylated Cell-Penetrating Peptide Bearing a Transferrin Receptor-Targeting Sequence for Neuro-Targeted siRNA Delivery. Molecular Pharmaceutics, 11 (2), 486–495. https://doi.org/10.1021/mp400446v

Zak MM⁻, Zangi L, Kowalski P, Boros G, Karikó K (2021) Lipid Nanoparticles for Organ-Specific mRNA Therapeutic Delivery. Pharmaceutics 2021, Vol. 13, Page 1675, 13 (10), 1675. https://doi.org/10.3390/PHARMACEUTICS13101675

Zhang H, Demirer GS, Zhang H, Ye T, Goh NS, Aditham AJ, Cunningham FJ, Fan C, Landry MP (2019) DNA nanostructures coordinate gene silencing in mature plants. Proceedings of the National Academy of Sciences of the United States of America, 116 (15), 7543-7548. https://doi.org/10.1073/PNAS.1818290116/SUPPL FILE/PNAS.1818290116.SM02.MP4

Zhang H, Goh NS, Wang JW, Pinals RL, González-Grandío E, Demirer GS, Butrus S, Fakra SC, del Rio Flores A, Zhai R, Zhao B, Park SJ, Landry MP (2021) Nanoparticle cellular internalization is not required for RNA delivery to mature plant leaves. Nature Nanotechnology 2021 17:2, 17 (2), 197-205.

https://doi.org/10.1038/s41565-021-01018-8

Zhang Y, Lai BS, Juhas M (2019) Recent Advances in Aptamer Discovery and Applications. Molecules, 24 (5). https://doi.org/10.3390/MOLECULES24050941

Zhu G, Chen X (2018) Aptamer-based targeted therapy. Advanced Drug Delivery Reviews, 134, 65. https://doi.org/10.1016/J.ADDR.2018.08.005

Zhu KY, Palli SR (2020) Mechanisms, Applications, and Challenges of Insect RNA Interference. Annual Review of Entomology, 65, 293. https://doi.org/10.1146/ANNUREV-ENTO-011019-025224

https://doi.org/10.1016/J.MOLMED.2022.04.007

Zhu S, Li W, Liu J, Chen CH, Liao Q, Xu P, Xu H, Xiao T, Cao Z, Peng J, Yuan P, Brown M, Liu XS, Wei W (2016) Genome-scale deletion screening of human long non-coding RNAs using a paired-guide RNA CRISPR library. Nature Biotechnology, 34 (12), 1279. https://doi.org/10.1038/NBT.3715

Zhu Y, Zhu L, Wang X, Jin H (2022) **RNA-based therapeutics: an** overview and prospectus. Cell Death & Disease 2022 13:7, 13 (7), 1–15. https://doi.org/10.1038/s41419-022-05075-2

Zotti M, dos Santos EA, Cagliari D, Christiaens O, Taning CNT, Smagghe G (2018) **RNA interference technology in crop protection against arthropod pests, pathogens and nematodes.** Pest Management Science, 74 (6), 1239–1250. https://doi.org/10.1002/PS.4813

Zou X, Wu J, Gu J, Shen L, Mao L (2019) Application of aptamers in virus detection and antiviral therapy. Frontiers in Microbiology, 10 (JULY), 1462.

https://doi.org/https://doi.org/10.3389/fmicb.2019.01462

Glossary

Antigen	Protein (or part thereof) of a pathogen that activates the immune system.
Antibody	Protein that binds specifically to a certain target molecule ($ ightarrow$ antigen). They have an important function in the immune response of vertebrates.
Antisense oligonucleotide (ASO)	Short, single-stranded RNA sequence with a length of 12–24 nucleotides. ASOs are produced synthetically to regulate gene expression as pharmaceuticals. ASOs bind to their target RNA and thereby influence it in different ways.
Arthropods	Arthropods are a phylum of the animal kingdom. Arthropods include insects, millipedes, crustaceans and arachnids.
Aptamer	Single-stranded RNA or DNA oligonucleotide or peptide that can bind to other molecules due to its three-dimensional structure.
CRISPR/Cas	Genome editing tool used to modify DNA at a specific location. The CRISPR/Cas complex consists of RNA molecules and a protein (enzyme).
Denaturation	Loss of the functional structure of a molecule, e.g. at high temperatures.
DNA	'Deoxyribonucleic acid': a macromolecule consisting of a chain of deoxyribonucleotides (adenine, guanine, cytosine and thymine). In all organisms, genetic information is present in the form of DNA.
dsRNA	'Double-stranded RNA': naturally occurring dsRNAs play an important role in RNA interference and thus gene regulation in almost all eukaryotes. Synthetically produced dsRNAs for gene regulation are developed or utilised as pharmaceuticals (in the form of → siRNA) or for agricultural applications.
Enantiomer	Enantiomers are molecules that have the same two-dimensional structure but are mirror images of each other in their spatial structure.
Eukaryotes	Living organisms whose cells have a nucleus, especially animals, plants and fungi.
Exon	Coding section of the mRNA that is retained during splicing, while \rightarrow introns are cut out of the pre-mRNA.
Genome editing	Process for the targeted modification of genetic material at a specific location. The most common genome editing tools include CRISPR/Cas, TALEN and zinc finger nucleases.
Host-induced gene silencing (HIGS)	A form of post-transcriptional gene silencing in plants. The DNA sequence for a \rightarrow dsRNA is inserted into the plant genome, resulting in a transgenic plant. The plant cells then express the corresponding dsRNA themselves (endogenously).
Intron	Section of the $ ightarrow$ pre-mRNA that is cut out during splicing – in contrast to exons – and is not contained in the mRNA.
Liposome	Spherical particle with a shell consisting of a double layer of lipids. Liposomes can be used to enclose molecules such as RNA and transport them into the cell interior.
LNAs	'Locked nucleic acids': nucleotides in which the sugar molecule has been chemically modified to increase stability.
miRNA	'microRNA': short, double-stranded RNA sequence with a length of 17–24 nucleotides, which can naturally regulate gene expression through RNA interference.
mRNA	'Messenger RNA': RNA that carries the genetic information for the construction of a specific protein.

Non-coding RNA (ncRNA)	RNA that contains no genetic information for the production of a protein.
Nuclease	Enzyme that breaks down nucleic acids by catalysing the hydrolysis of the bonds between the individual nucleotides.
Nucleoside	Compound of a nucleobase and a sugar molecule. Phosphorylation turns the nucleoside into a nucleotide, the basic building block of nucleic acids.
Nucleotide	Basic chemical building block of nucleic acid. A nucleotide consists of one of four specific nucleic bases, a sugar molecule and at least one phosphate group.
Off-target effect	Unintended effect outside the target sequence.
Oligonucleotide	Short DNA or RNA sequence of a few nucleotides (usually up to 10–200 nucleotides).
Phosphorothioate (PS)	In order to stabilise RNAs, the natural phosphodiester bonds between the nucleotides are replaced by synthetic phosphorothioate bonds, particularly in the case of \rightarrow ASOs.
Polyethylene glycol (PEG)	PEG is a water-soluble polymer that has many applications in medicine, research and industry. It is usually categorised as non-toxic but can trigger allergic reactions.
pre-mRNA	Preform of mRNA that is produced directly after transcription. The pre-mRNA is modified in various ways and further processed into \rightarrow mRNA. The modifications include the addition of the poly(A) tail and the cap structure at both ends of the pre-mRNA as well as splicing.
pre-miRNA (precursory miRNA)	Preform of \rightarrow miRNA, which is created by processing \rightarrow pri-miRNA. (see pri-miRNA).
Primer	Sequence that serves as the starting point for DNA replication.
pri-miRNA	Preform of miRNA that is produced during transcription. It is significantly longer than the miRNA and is bundled together in a loop.
Ribosome	Complex of RNA and proteins that catalyses the translation and thus the production of proteins in the cells.
RNA	'Ribonucleic acid': a macromolecule consisting of a chain of nucleotides. In contrast to DNA, RNA contains the base uracil instead of thymine and has an additional hydroxyl group on the sugar-phosphate backbone.
RNA activation (RNAa)	Method for increasing the transcription of a gene.
RNA aptamer	Short, non-coding, single-stranded RNA that binds specifically to a target molecule due to its three-dimensional shape.
RNA interference (RNAi)	Mechanism for regulating gene expression that occurs naturally in almost all eukaryotic organisms. In RNAi, small RNA sequences bind to their complementary target RNA and inhibit the production of the corresponding protein.
saRNA	'Small activating RNA': short, chemically synthesised, double-stranded RNA that binds specifi- cally to a target sequence in the genome and thereby causes increased transcription via 'RNA activation'.
shRNA	'Short hairpin RNA': short, double-stranded RNA with a hairpin-shaped structure. shRNAs are processed into \rightarrow siRNAs in cells and thus play a role in RNA interference.
siRNA	'small interfering RNA': short, double-stranded RNA sequence with a length of 21–24 nucleotides, which can regulate gene expression both naturally through $→$ RNA interference and in synthetic form as pharmaceuticals. siRNAs bind complementary target RNA and thereby inhibit it.
Spiegelmers	Spiegelmers are enantiomers of aptamers.
Splicing	Removal of certain sequences (\rightarrow introns) from the pre-mRNA. Splicing is an important step in the further processing of pre-mRNA into mRNA. If different sequences are cut from the pre-mRNA by splicing, different mRNAs and thus different proteins can be produced.

Spray-induced gene silencing (SIGS)	A form of \rightarrow gene silencing in plants. The dsRNA is sprayed onto the plant from the outside (exogenously) using a spray.
sRNA	'small RNA': short, non-coding RNA with an approximate length of less than 200 nucleotides.
Transcript	RNA sequence read (transcribed) from the genetic information.
Transcription	Overwriting the DNA in \rightarrow pre-mRNA, the preform of mRNA.
Translation	Translation of mRNA into amino acids and thus proteins.

SCNAT - network of knowledge for the benefit of society

The **Swiss Academy of Sciences (SCNAT)** works at regional, national and international level for the future of science and society. It strengthens the awareness for the sciences as a central pillar of cultural and economic development. The breadth of its support makes it a representative partner for politics. The SCNAT links the sciences, provides expertise, promotes the dialogue between science and society, identifies and evaluates scientific developments and lays the foundation for the next generation of natural scientists. It is part of the association of the Swiss Academies of Arts and Sciences.

The **Genetic Research Forum** of the Swiss Academy of Sciences (SCNAT) focusses on developments in genetic research and their impact on society. The forum promotes dialogue between scientists, decision-makers and the public.

